

Trade-Offs for Climate-Smart Forestry in Europe Under Uncertain Future Climate

Konstantin Gregor, Thomas Knoke, Andreas Krause, Christopher Reyer, Mats Lindeskog, Phillip Papastefanou, Benjamin Smith, Anne-sofie Lansø, Anja Rammig

▶ To cite this version:

Konstantin Gregor, Thomas Knoke, Andreas Krause, Christopher Reyer, Mats Lindeskog, et al.. Trade-Offs for Climate-Smart Forestry in Europe Under Uncertain Future Climate. Earth's Future, 2022, 10 (9), 10.1029/2022EF002796 . hal-03775325

HAL Id: hal-03775325 https://hal.science/hal-03775325v1

Submitted on 12 Sep 2022

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.





RESEARCH ARTICLE

10.1029/2022EF002796

Key Points:

- Strategies for climate-smart forestry under a range of climate scenarios always lead to trade-offs between different ecosystem services (ESs)
- Higher shares of unmanaged and broad-leaved forests are beneficial for numerous ESs, but lead to decreased timber provision
- The mitigation potential of forests strongly relies on substitution effects which depend on the carbon-intensity of the alternative products

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

K. Gregor, konstantin.gregor@tum.de

Citation:

Gregor, K., Knoke, T., Krause, A., Reyer, C. P. O., Lindeskog, M., Papastefanou, P., et al. (2022). Trade-offs for climate-smart forestry in Europe under uncertain future climate. *Earth's Future*, *10*, e2022EF002796. https://doi. org/10.1029/2022EF002796

Received 23 MAR 2022 Accepted 24 AUG 2022

© 2022. The Authors. Earth's Future published by Wiley Periodicals LLC on behalf of American Geophysical Union. This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Trade-Offs for Climate-Smart Forestry in Europe Under Uncertain Future Climate

Konstantin Gregor¹, Thomas Knoke¹, Andreas Krause¹, Christopher P. O. Reyer², Mats Lindeskog³, Phillip Papastefanou^{1,3}, Benjamin Smith^{3,4}, Anne-Sofie Lansø^{5,6}, and Anja Rammig¹

¹TUM School of Life Sciences, Technical University of Munich, Freising, Germany, ²Potsdam Institute for Climate Impact Research, Member of the Leibniz Association, Potsdam, Germany, ³Department of Physical Geography and Ecosystem Science, Lund University, Lund, Sweden, ⁴Hawkesbury Institute for the Environment, Western Sydney University, Penrith, NSW, Australia, ⁵Department of Environmental Science, Aarhus University, Aarhus, Denmark, ⁶Laboratoire des Sciences du Climat et de l'Environnement (LSCE/IPSL) CEA-CNRS-UVSQ, Université Paris-Saclay, Gif-sur-Yvette, France

Abstract Forests mitigate climate change by storing carbon and reducing emissions via substitution effects of wood products. Additionally, they provide many other important ecosystem services (ESs), but are vulnerable to climate change; therefore, adaptation is necessary. Climate-smart forestry combines mitigation with adaptation, whilst facilitating the provision of many ESs. This is particularly challenging due to large uncertainties about future climate. Here, we combined ecosystem modeling with robust multi-criteria optimization to assess how the provision of various ESs (climate change mitigation, timber provision, local cooling, water availability, and biodiversity habitat) can be guaranteed under a broad range of climate futures across Europe. Our optimized portfolios contain 29% unmanaged forests, and implicate a successive conversion of 34% of coniferous to broad-leaved forests (11% vice versa). Coppices practically vanish from Southern Europe, mainly due to their high water requirement. We find the high shares of unmanaged forests necessary to keep European forests a carbon sink while broad-leaved and unmanaged forests contribute to local cooling through biogeophysical effects. Unmanaged forests also pose the largest benefit for biodiversity habitat. However, the increased shares of unmanaged and broad-leaved forests lead to reductions in harvests. This raises the question of how to meet increasing wood demands without transferring ecological impacts elsewhere or enhancing the dependence on more carbon-intensive industries. Furthermore, the mitigation potential of forests depends on assumptions about the decarbonization of other industries and is consequently crucially dependent on the emission scenario. Our findings highlight that trade-offs must be assessed when developing concrete strategies for climate-smart forestry.

Plain Language Summary Forests help mitigate climate change by storing carbon and via avoided emissions when wood products replace more carbon-intensive materials. At the same time, forests provide many other "ecosystem services (ESs)" to society. For example, they provide timber, habitat for various species, and they cool their surrounding regions. They are, however, also vulnerable to ongoing climate change. Forest management must consider all these aspects, which is particularly challenging considering the uncertainty about future climate. Here, we propose how this may be tackled by computing optimized forest management portfolios for Europe for a broad range of future climate pathways. Our results show that changes to forest composition are necessary. In particular, increased shares of unmanaged and broad-leaved forests are beneficial for numerous ESs. However, these increased shares also lead to decreases in harvest rates, posing a conflict between wood supply and demand. We further show that the mitigation potential of forests strongly depends on how carbon-intensive the replaced materials are. Consequently, should these materials become "greener" due to new technologies, the importance of wood products in terms of climate change mitigation decreases. Our study highlights that we cannot optimize every aspect, but that trade-offs between ESs need to be made.

1. Introduction

Forests contribute to climate change mitigation (Canadell & Raupach, 2008; Pan et al., 2011; Ramstein et al., 2019) through their significant role in the global carbon cycle. Annually, over 25% of global anthropogenic carbon emissions are absorbed by the terrestrial biosphere, primarily by forests (Friedlingstein et al., 2020; Harris et al., 2021; Pan et al., 2011). Consequently, Article 5 of the Paris Agreement states specifically that the world's

existing forest carbon sinks should be conserved or enhanced (United Nations, 2015a). So far, this remains a challenging task since forest carbon sinks are vulnerable to climate change, mainly due to increased frequency and severity of disturbances (Dai, 2013; IPCC, 2014; Seidl et al., 2014; Spinoni et al., 2018).

European forests are estimated to take up 9% of Europe's emissions (Grassi et al., 2019). Since 90% of these forests are managed (IPCC, 2014), they additionally provide considerable amounts of wood products that store carbon and substitute carbon-intensive materials and fuels (Grassi et al., 2021; Harmon, 2019; Howard et al., 2021; van Kooten & Johnston, 2016). Furthermore, apart from climate change mitigation, forests provide numerous other ecosystem services (ESs) such as, for example, timber production, water regulation, local climate regulation, and recreation (Binder et al., 2017; Cordonnier et al., 2014; Mori et al., 2017). Therefore, future management strategies must also consider the continued provision of these manifold ESs (Brockerhoff et al., 2017; Díaz et al., 2006; Hua et al., 2022; Millenium Ecosystem Assessment, 2005), which is also reflected in the UN's sustainable development goals (United Nations, 2015b) and the declaration of sustainable forest management in Europe (Forest Europe, 2015a). This holistic approach of jointly considering adaptation, mitigation, and ESs in forest management has been coined "climate-smart forestry" (Kauppi et al., 2018; Nabuurs et al., 2017).

One approach to tackle the issue of assessing multiple goals in forest planning is using multi-criteria decision making (MCDM, Ishizaka and Nemery [2013], see also Uhde et al. [2015]). MCDM has been applied to find the best possible forest management regarding multiple ESs under different climate scenarios (e.g., in Spain, Diaz-Balteiro et al., 2017). Using models to develop and assess forest management portfolios—defined as the relative proportions of species and forest management alternatives—allows for a more nuanced assessment since these models integrate several ESs. In addition, portfolios can combine the benefits of multiple management options. This was recently demonstrated by Luyssaert et al. (2018) who created forest management portfolios from six simplified management options (e.g., species changes, conversion to coppice, or refraining from management) and applied these across Europe using the ORCHIDEE-CAN model. However, their optimization of three single-criterion objectives (maximize albedo, maximize carbon sink, minimize surface temperature) entailed trade-offs that must be made between the optimized objective and other ESs.

Management decisions in forestry are particularly complicated: The long life-spans of trees and changing societal demands require assumptions about the future resulting in large uncertainties. These uncertainties may be addressed with the concept of robust optimization (Ben-Tal & Nemirovski, 2002; Gorissen et al., 2015). One of the first studies that applied robust optimization to land-use portfolios was done by Knoke et al. (2015). Furthermore, several recent studies have shown how robust optimization and MCDM can be combined to optimize forest management portfolios for multiple ESs considering uncertainty (Knoke et al., 2016, 2020; Uhde et al., 2017).

Currently, one of the most uncertain aspects of forest planning is climate change and its local impacts (Lindner et al., 2014). For instance, it is challenging to find tree species that are well-suited for present and future conditions (Hickler et al., 2012). Such future uncertainty can be explored in modeling studies where various species can be assessed under different climate scenarios based on Representative Concentration Pathways (RCPs) covering several plausible trajectories of anthropogenic emissions and climates (Dufresne et al., 2013; IPCC, 2018; Meinshausen et al., 2011). However, recent studies on future forest management strategies do not embrace the full range of uncertainty; therefore ignoring the issue that a proposed forest management option deemed practical for one RCP might not be useful if another RCP materializes. Alternatively, it is crucial to broadly evaluate the full range of possible climates (Pedersen et al., 2020) and provide solutions that will be practical across all RCPs. This prevents costly adaptation if the world develops in unforeseen ways (Lawrence et al., 2020) and is also promoted by the IPCC which states that also low-likelihood outcomes need to be included in risk assessments (IPCC, 2021).

To address this, we considered RCP2.6, RCP4.5, RCP6.0, and RCP8.5, and determined how forest management portfolios may be constructed to provide all ESs across these RCPs. We combined the well-established dynamic vegetation model LPJ-GUESS (B. Smith et al., 2001; B. Smith et al., 2014; Lindeskog et al., 2021) with MCDM and robust optimization to consider the ESs climate change mitigation, timber provision, water availability, local climate regulation, and biodiversity habitat. Additionally, we covered climate change adaptation by evaluating which forest types can survive the different climate scenarios. Specifically, we focused on European forests, using six simplified forest management options similar to Luyssaert et al. (2018). Our approach allowed us to assess several ESs simultaneously under uncertain future climate development across Europe. We present a set of



management portfolios for different European forest types that are viable across many future climate scenarios. We further investigated the potential trade-offs arising when implementing such portfolios, mainly by considering harvest rates and mitigation impacts. Lastly, we evaluated how different preferences and assumptions affected the portfolios.

2. Methods

We applied LPJ-GUESS for the European domain with a $0.5^{\circ} \times 0.5^{\circ}$ spatial resolution and combined it with an optimization framework (Figure 1). We performed 24 simulations for m = 6 management options and n = 4 RCPs (RCP2.6, RCP4.5, RCP6.0, RCP8.5). We computed seven ES indicators (ESIs—representing the ESs climate change mitigation, local climate regulation, biodiversity habitat, timber provision, and water availability) to evaluate the different management options (Table 2). Finally, our optimization algorithm computed one management portfolio per grid cell which allowed an optimally balanced provision of all considered ESIs across all RCPs.

2.1. Dynamic Vegetation Model and Simulation Protocol

LPJ-GUESS (B. Smith et al., 2001; B. Smith et al., 2014) is a dynamic vegetation model that simulates establishment, growth, competition, management, and mortality of plant functional types (PFTs). Each PFT is represented by different parameters governing, for example, life-history strategy, phenology, growth form, drought tolerance, bioclimatic limits, and others (B. Smith et al., 2014). The modeled processes include photosynthesis and stomatal conductance based on BIOME3 (Haxeltine & Prentice, 1996), stochastic implementations of population dynamics (B. Smith et al., 2001), allocation of carbon to different compartments and allometric relationships for plant growth (Sitch et al., 2003). They also include the nitrogen cycle, soil, and litter processes (B. Smith et al., 2014). LPJ-GUESS has been designed to simulate the vegetation response under future climate change, including responses to climate extremes. As such it has been benchmarked to a wide variety of independent datasets (e.g., Chang et al., 2017; Haverd et al., 2020; Ito et al., 2017; Lindeskog et al., 2021; B. Smith et al., 2014).

Mortality is implemented as a stochastic process considering a tree's growth efficiency and age. The simulations in this study were carried out in "cohort-mode," where one average individual represents a cohort of the same age. Each stand was modeled via 25 replicate patches representing random samples of the same stand.

Disturbances in LPJ-GUESS are modeled stochastically as patch-destroying disturbances. Our study followed Pugh et al. (2019) and used different average return times for different stand types. We used their global median values of 1000 and 500 years for broad-leaved deciduous and evergreen stands, respectively, and 300 years for needle-leaved and mixed stands. Note that we used these values even though they contain management disturbances, since return times in Europe are partly much lower than these global median values (Pugh et al., 2019). Additionally, we simulated fire using the GlobFIRM fire model (Thonicke et al., 2001) in the historical and spin-up phase to retrieve a reasonable representation of present-day European forests (see Section 3.1.1); however,



Figure 1. Visualization of the methodology, leading to one single portfolio per grid cell. Panel (a) $n \times m$ independent model simulations are run for the management options and Representative Concentration Pathways (RCPs). (b) Ecosystem service indicators (ESIs)—if not computed by LPJ-GUESS directly—are computed from the model outputs, then aggregated (e.g., 2100–2130 mean), and normalized to [0; 1], resulting in one table per grid cell containing the normalized values for all management types and RCPs. (c) For each grid cell, one optimization is run resulting in (d) one optimized portfolio per grid cell which for any RCP ensures a balanced provision of all ESIs. (e) Additional parameters may be passed into the process: A parameter $\lambda \in [0, 1]$ governs the focus on the balanced provision of the ESI. A low λ results in a more balanced provision of all ESIs whereas a high λ lays more focus on the average ESI values (and less balance) instead, see also Section 2.3. Weights can be added to put more focus on one or more ESI. Additional constraints can put bounds on the variables, for example, harvest needs to be at least at 2010 levels.

we excluded fire for the future period as it was included in the disturbance intervals. Furthermore, starting in 2010, we increased the disturbance rates by 1% each year to account for expected increasing disturbance frequencies in Europe (Senf & Seidl, 2021a).

We used specific parametrizations for 18 European tree species and four more generic PFTs for three shrubs and grass (Table S1 in Supporting Information S1). We used the parameters of LPJ-GUESS v4.0 (based on the values proposed by Hickler et al. [2012]) with adaptations as specified in Lindeskog et al. (2021). However, we removed the maximum temperature of the coldest month bioclimatic limit for establishment (tcmax_{est}) of certain species (as in Hickler et al. [2012]) after observing that otherwise modeled present-day occurrences were too far off observations for these species (Mauri et al., 2017).

2.1.1. Representation of Forest Management in LPJ-GUESS

Several forest management options are implemented in LPJ-GUESS (Lindeskog et al., 2021). Here we applied the thinning scheme following the self-thinning rule of Reineke (1933), which was implemented similarly as in Bellassen et al. (2010) as described in Lindeskog et al. (2021). Apart from thinning, forests were clear-cut when a stand fell below a corresponding target density predefined for broad-leaved summer green and needle-leaved forests (Bellassen et al., 2010; Lindeskog et al., 2021). We defined an additional target density for broad-leaved evergreen forests (75 ind/ha) to yield comparable rotation times between the forest types to facilitate comparison of harvests between forests in the limited time period of interest (2100–2130). In achieving the correct age distribution by 2010, clear-cuts were omitted between stand establishment and 2010 (see Section 2.1.3). For stands that attained the target density before 2010, clear-cuts were spread out through 2010–2020 to avoid a significant anomaly in 2010 (see Figure S20 in Supporting Information S1). During harvesting, 65% of the total carbon of the sapwood and heartwood was considered stem material of which 90% ("harvest efficiency") were taken out. Harvests were distributed to different product pools as described in Section 2.1.4. Small fractions of leaves and branches were assumed to be taken out and burned, and the rest was left to decay on-site together with the stem residues and coarse roots (see Lindeskog et al., 2021). We also implemented salvage logging, described in detail in Section S1.1 of Supporting Information S1.

2.1.2. Coppice Management Implementation in LPJ-GUESS

Coppice is a forest type that historically was very important in Europe (Albert & Ammer, 2012; Evans, 1984), still accounting for large forest shares in Southern European countries (Maganotti et al., 2018). Therefore, we implemented a simple form of coppice management into LPJ-GUESS similar to ORCHIDEE_CAN_r3069 (Luyssaert et al., 2018). If a broad-leaved tree is cut down, its coarse roots (assumed to be 22% of the total carbon mass) stay alive, and new shoots emerge from the stump (assumption: 10 per stump for all species). Choosing the initial number of shoots after a coppice event showed no significant difference after a few years as the mortality routines of LPJ-GUESS consequently resulted in strong self-thinning in cases with a higher number of shoots (see Figure S4 in Supporting Information S1). Besides this expected self-thinning (Johansson, 2008; Leonardsson & Götmark, 2015; Rydberg, 2000; Verlinden et al., 2015) we also applied anthropogenic thinning in coppices (Nicolescu et al., 2017). The shoots were harvested individually once they reached a diameter of 20 cm, our coppice forests were thus representative of traditional coppice forests, not short-rotation coppice plantations.

2.1.3. Simulation Protocol for LPJ-GUESS Simulations

For our simulations we followed the setup of Lindeskog et al. (2021) by using monthly temperature, radiation, and precipitation data (including the number of wet days) from CMIP5 (Taylor et al., 2012) simulations of the general circulation model (GCM) IPSL-CM5A-MR (Dufresne et al., 2013). We selected this GCM to compare our results with Luyssaert et al. (2018).

The climate data was interpolated bi-linearly from $2.5^{\circ} \times 1.25^{\circ}$ to $0.5^{\circ} \times 0.5^{\circ}$ spatial resolution and bias-corrected against 1961–1990 observations from CRU-NCEP (see Ahlström et al., 2012). For each climatic variable, this data covered a transient "historical" (1850–2005) and a future period (2006–2100) per RCP.

After a 1200-year spin-up period to bring soil carbon pools close to equilibrium, the simulations were run from 1800 until 2130. Before 1850, the 1850–1879 climate was detrended and recycled. It was similarly performed with each scenario's 2071–2100 data for the time after 2100. Our optimization considered the period 2100–2130. By then, most simulated forests had undergone a full rotation (Figure S22 in Supporting Information S1).

 CO_2 concentrations of the RCPs were taken from Meinshausen et al. (2011), and a constant pre-industrial (year 1765) atmospheric CO_2 concentration of 278 ppmv was applied for the spin-up. Decadal values for Nitrogen deposition from 1850 to 2100 were used, depending on the grid cell and RCP (Lamarque et al., 2011).

To commence projections from a realistic representation of forests in Europe today, forest stands established during spin-up were successively clear-cut and replaced by stands matching the species distribution map from Brus et al. (2012). The timing of these clear-cuts was predefined to reproduce the observed 2010 age distribution from the global forest age data set (GFADv1.0, Poulter et al., 2018). This is why we turned off disturbances and clear-cuts in such stands between their establishment and 2010.

Old-growth forests (i.e., older than 140 years in 2010 according to GFAD) were not subject to any management before and after 2010. We kept the forest area constant at 2010 levels for future projections and focused on "forests remaining forests." Large-scale deforestation is unlikely due to the Paris Agreement (United Nations, 2015a) and national laws. Forest area increase however has recently been observed (Forest Europe, 2020). This trend might continue if land abandonment outweighs effects of increasing demand for agricultural products (Luyssaert et al., 2018; Perpiña Castillo et al., 2021; P. Smith et al., 2010). This competition is not part of our study.

We performed one simulation until 2010 for all 3124 grid cells in Europe to evaluate the present-day model outputs against observations and other models (Section 3.1.1). To create forest management portfolios, we implemented six simplified forest management options (Table 1) based on Luyssaert et al. (2018). We performed one future projection simulation (2010–2130) for each RCP and management option. For example, in the toBd ("to broad-leaved deciduous") simulation, all stands were converted to a broad-leaved deciduous forest after clear-cut. When changing a non-broad-leaved stand to coppice, the stand was initially converted to a broad-leaved deciduous outputs of the managed as coppice.

We sampled grid cells every 2° latitude and longitude for future runs to reduce the computational load resulting in simulations for 193 grid cells across Europe. In the optimization, we combined the management options to create a portfolio for each grid cell since the stands within a grid cell were independent from one another. We excluded cells from the portfolio analysis where the average harvests between 1990 and 2010 were below 0.1 gC/m²/yr or less than 25% of the forest area was converted by 2100. The reason for this is that with such a low conversion, the forests looked very similar at the end of the simulation period and the difference in ESIs were only marginal, not allowing for a sensible comparison of management options. This resulted in a final set of 181 grid cells.

2.1.4. Contributions of Forests to Climate Change Mitigation

Apart from ecosystem carbon storage, the effect of forests in mitigating climate change depends on the life cycle of wood products, attributed to their ability to substitute carbon-intensive materials and fuels (Grassi et al., 2021; Harmon, 2019; Howard et al., 2021; van Kooten & Johnston, 2016). To measure this, we followed Krause et al. (2020) by using two wood product pools for medium and long-lived products, respectively, with corresponding decay rates described by two Gamma-functions (Klein et al., 2013, Figure S24 in Supporting Information S1).

Table 1 The Six Simplified Management Options				
Name	Explanation			
base	After clearcut, plant the same species as before, same management as before (thinning and clearcut)			
toBd	After clearcut, convert to broad-leaved deciduous forest by planting the most common broad-leaved deciduous species in the stand. Continue with thinning and clearcut			
toBe	Like toBd, but planting Q. ilex, the LPJ-GUESS broad-leaved evergreen species			
toCoppice	Broad-leaved forests are clearcut and left to regrow from the stumps, needle-leaved forests are clearcut and replaced with broad-leaved species. The forest is managed as coppice from then on			
toNe	Like toBd, but planting the most common needle-leaved evergreen species in the stand			
unmanaged	Refrain from the clearcut and also refrain from any thinnings after this point in time, leaving the forests completely untouched			
Note A monogom	ant desision was taken for each stand ofter 2010 as soon as it reached maturity (i.e. a target density). The			

Note. A management decision was taken for each stand after 2010 as soon as it reached maturity (i.e., a target density). The conversion was implemented by planting the most common species of each forest type for that grid cell.

These functions account for the fact that the entire amount of wood entering a product pool will typically remain there for some time before it starts to be returned to the atmosphere.

We based product flows on Klein et al. (2013) and Eurostat (2021a), where for conifers (non-conifers), 23% (2.5%) of stem mass entered the long-lived pool, and 9.4% (11.9%) entered the medium pool. Meanwhile, the rest (plus some harvest residues) was converted to fuel wood or short-lived products and transferred to the atmosphere within one year (Figure S19 in Supporting Information S1). For coppice, we assumed that 2% entered the medium product pool. The rest was turned over immediately, because most wood from coppice forests is used for energy, heating, and paper, and only very small portions for, for example, furniture (Maganotti et al., 2018). For the medium and long-lived pools the median residence times were 18 and 93 years, respectively, that is, 50% of products remained in that pool after this time (Figure S24 in Supporting Information S1).

To assess the emission reductions, we used displacement factors (defined as avoided emissions in relation to the mass of carbon in the wood product, Arehart et al., 2021) for fuels and materials, respectively. We used 0.67 tC/tC as the displacement factor for fuels (Knauf et al., 2015). It is based on a simple assessment of the possible energy provision of wood compared to light oil (Rüter, 2011). It is comparable to those of studies of other developed nations (Myllyviita et al., 2021) and well in the range of "less than 0.5 up to about 1.0" as suggested by Sathre and O'Connor (2010). The displacement factor for material substitution is based on an analysis of 16 key products and their alternatives, making up 90% of the wood usage spectrum (Knauf et al., 2015). Its value of 1.5 tC/tC was originally estimated for Germany and is lower than the commonly used value of 2.1 (Sathre & O'Connor, 2010) which is a mean value over studies considering different applications and countries across the world. It does not contain end-of-life assessment. We used this displacement factor for 77% of wood products and assumed energy recovery at the end of their lifetime. In Europe, 23% of all waste ends up in landfills (Eurostat, 2021b). Consequently, for the remaining 23% of products we used a lower displacement factor of 1.1 tC/tC to also account for emissions from landfills (Sathre & O'Connor, 2010).

Wood fuels emit more CO_2 per unit of energy than fossil fuels, and it requires a long time until a forest has regrown and absorbed the initially emitted carbon ("carbon debt," Cherubini et al., 2011). Consequently, wood-based fuels are not carbon-neutral, at least on short and medium time-scales (Booth et al., 2020; Cherubini et al., 2011; Holtsmark, 2012; Leturcq, 2020). It is crucial to take this into account, which is done here by considering the total carbon effect including stocks in forests and products, similarly as performed by Knauf et al. (2016). This approach quantifies the actual emissions from burning the wood.

The actual number of avoided emissions heavily depends on the energy mix at a given time or the type of materials that are replaced and is expected to decrease in the future (Harmon, 2019). We accounted for this by gradually discounting the displacement factors based on the RCPs, similar to Brunet-Navarro et al. (2021, Figure S23 in Supporting Information S1). Additionally, we evaluated our approach without discounting (see Section 4.3).

Finally, it is crucial to take into account additionality. Any measurement of mitigation from forests needs to be compared to a reasonable baseline, because only an additional measure would contribute to climate change mitigation. In our case this is done implicitly by including a baseline simulation ("base"). Any change in forestry can thus be compared to this baseline.

2.2. Ecosystem Services and Their Indicators

Our study investigated the potential changes in seven ESIs (Table 2 and Section S1.2 in Supporting Information S1). As indicators for the provision of timber we used both the simulated total harvest values ($tha^{-1} yr^{-1} dry$ mass) and harvests used only for long-lived wood products (HLP).

We used one combined indicator for a forest's (global) climate change mitigation potential by combining the total simulated carbon pool (vegetation + soil + litter + products) with cumulative avoided emissions through material and energy substitution. For local climate regulation we included evapotranspiration (ET) and surface roughness. Since the *local* effects of albedo have previously been found to be small compared to ET and surface roughness (Winckler et al., 2019), we included albedo only in an additional experiment. Although we acknowledge that biogeophysical variables also possess non-local and global effects (Pongratz et al., 2010; Winckler et al., 2017), these might be small compared to the biogeochemical effects (Pongratz et al., 2010). Therefore, we solely evaluated increases in these variables as an indication for local cooling, omitting possible feedback on global

Variable Name	Ecosystem service indicator (ESI)	Explanation		
Harvests	Total harvests	Total wood provision (including products like firewood, pulp etc.)		
HLP	Harvests for long-lived wood products	Timber provision for long-lived products (e.g. construction)		
Mitigation	Carbon storage plus material and energy substitution effects	Total carbon in vegetation, soil, litter, and products, plus avoided emissions from substitution with wood products		
<i>z</i> ₀	Surface roughness	Indicator for atmospheric conductance, influencing heat fluxes. Higher roughness results in higher fluxes		
ET	Total evapotranspiration	Indicator for latent heat fluxes. More ET means more local cooling		
Ψ_{soil}	Soil water potential	Yearly minimum of monthly values, indicator of water availability and drought stress		
Bio	Combined indicator (RCP-normalized mean) of amount of coarse woody debris, Shannon entropy of 5 cm diameter-at-breast-height (DBH) classes and number of trees with DBH >50 cm	Coarse woody debris, large trees, and an abundance of various tree sizes provide high numbers of habitats and resources (Cordonnier et al., 2014)		

Table 2

Note. Albedo was included in an additional experiment. See Section \$1.2 in Supporting Information \$1 for technical details.

climate. The yearly minimum of monthly soil water potential Ψ_{soil} was used as an indicator for water availability (Rajasekaran et al., 2018), where a decrease in Ψ_{soil} indicates less water available for plants. We implemented a combined indicator for biodiversity based on tree sizes and dead wood, adapted from Cordonnier et al. (2014). Due to the model setup (i.e., planting the most common species of the grid cell), it was not sensible to include an indicator for the diversity of tree species. Finally, to assess the changes in ESI performances, we compared their values at the beginning of the next century (mean of 2100–2130) to present-day values (mean of 2000–2010).

2.3. Robust Multi-Criteria Optimization

To compute portfolios that provide various ESs of different units and magnitudes, we used MCDM (e.g., Ishizaka & Nemery, 2013; Marler & Arora, 2004). To account for uncertainty in the inputs stemming from the RCPs, we combined this with robust optimization (Ben-Tal & Nemirovski, 2002). The solution of such an optimization guarantees feasibility for all realizations of the uncertainty set and can be written as a maxi-min linear program (Gorissen et al., 2015). To enable comparison, the computed ESIs from the simulations were first normalized to [0, 1]. This was done for each ESI and RCP separately, since ESI values of different RCPs do not belong to the same environmental conditions and we argue that they cannot be compared to an ESI from an entirely different climatic future. Our normalization thus yields a better interpretability of the worst case results: When the worst case ESI is, say, 0.4, this means that in every RCP we will achieve 40% of the achievable outcome in this emission pathway. For each grid cell this resulted in one linear program creating a single portfolio for all RCPs:

$$\max_{\omega} \left((1-\lambda) \min_{esi,rcp} \sum_{s} \omega_{s} W_{esi} q(esi, s, rcp) + \lambda \sum_{esi,rcp} \sum_{s} \omega_{s} W_{esi} q(esi, s, rcp) \right)$$
(1)

subject to
$$\sum_{s \in S} \omega_s = 1$$
 (2)

$$\omega_s \ge 0 \qquad \forall \ s \in S \tag{3}$$

$$\sum_{esi} W_{esi} = 1 \tag{4}$$



$$W_{esi} \ge 0 \quad \forall \ esi$$
 (5)

 $fpc(2100, s, rcp) \ge min(0.1, fpc(2010))$ (6)

where $S = \{$ base, toBd, toBe, toCoppice, toNe, unmanaged $\}$

 ω_s : Share of management type *s* in the optimized portfolio

 W_{esi} : Preference / weight for esi

fpc(year, s, rcp) : Foliar Projective Cover of the grid cell under management option s in RCP rcp in year year

q(esi, s, rcp): Normalized quality of esi for management option s in rcp

 $\sum_{s \in S} \omega_s q(esi, s, rcp) :$ Quality of *esi* for the whole grid cell for a portfolio ω (without ESI weights)

The objective function, Equation 1, expresses that we wanted to find the portfolio ω that results in optimally balanced ESIs over all RCPs. It is a weighted combination of the worst case ESI and the average ESI. Optimizing the worst case would lead to the most balanced solution but is quite pessimistic, since a bad performance of one ESI cannot be compensated with the excellent performance of another ESI. Meanwhile, optimizing the average ESI, results in a much less balanced solution but could improve some ESIs. In the theory of sustainability, the former approach termed "strong sustainability" (Ruiz et al., 2011) is preferred (Diaz-Balteiro et al., 2018). We combined the two approaches (Diaz-Balteiro et al., 2018) by adding a trade-off parameter $\lambda = 0.2$ that resulted in balanced ESI performances and robustness to small changes in ESIs, thus obtaining more similar results in adjacent grid cells. We argue that a higher λ -value is unreasonable since it would invalidate the approach of treating all RCPs and ESIs equally. Also, mathematically, a higher λ would decrease the relevance of the worst cases (left summand) due to their naturally lower value than the averages (right summand).

Equations 2 and 3 ensure that our portfolio contained non-negative fractions for all strategies and allotted the whole managed forest area of the grid cell. Additionally, weights can be given to each ESI (Equations 4 and 5). Throughout this study we applied equal preferences W_{esi} unless otherwise noted. Finally, Equation 6 ensures the maintenance of forest cover at the beginning of the next century, using a value of foliar projective cover of 10% based on FAO's definition of forests (FAO, 2020), or the 2010 foliar projective cover if lower than 10%. Notably, violation of this equation led to a management option not to be considered for a grid cell, since such a low foliar projective cover indicates that this type of forest does not grow well under the given conditions. This also led to the exclusion of this management option from the normalization process and the *q*-values for the other management options. The normalization was excluded from the equations above for brevity, and the optimization was implemented using SciPy (Virtanen et al., 2020).

3. Results

3.1. Simulation Results

3.1.1. Model Evaluation

We found that LPJ-GUESS reproduced present-day data from observations and models (Table 3, Figures S6–S8 in Supporting Information S1). For example, modeled vegetation measures such as tree cover, biomass, net and gross primary productivity (NPP and GPP), ET, and runoff were close to estimates from the literature, on average over the continent and more regionally. Total carbon was overestimated. Additionally, simulated fellings were above values from country reporting. We found a historical increase in GPP of 30% in response to changing environmental conditions such as increased CO_2 -concentrations from 1900 to 2010 (Figure S1 in Supporting Information S1).

3.1.2. Projected Changes in ESIs Until 2100-2130

In our simulations, a European forest landscape with solely unmanaged forest by 2100-2130 resulted in the highest carbon stocks (including products) with 12.1-12.4 kgC/m² (Tables S3–S6 in Supporting Information S1) over the 181 analyzed grid cells which is on average 19% above present-day values and 24% above the base scenario. Other management options generally resulted in carbon stocks below present-day values in all RCPs.

Table 3

Comparison of Our Present-Day (2000-2010 Average) Simulation Outputs to Results From Other Studies and Publicly Available Datasets

	This study	Literature	Data source ^a	Reference
Vegetation C (PgC)	13.7	13	Ι	Pan et al. (2011)
		11.7 ^b	С	Forest Europe (2015b)
		11.6°	S	Liu et al. (2015)
Total C (PgC)	59.8	40.9 ^d	Ι	Pan et al. (2011)
Forest-GPP (gC/yr/m ²)	1175	1107-1199	E, F	Luyssaert et al. (2010)
Forest-NPP (gC/yr/m ²)	426	447	Ι	Luyssaert et al. (2010)
Fellings (10 ⁶ m ³ yr ⁻¹)	663	562	С	Forest Europe (2015b)
Total Tree Cover (%)	24.9	26.8 ^e	S	GFC v1.7 Map (Hansen et al., 2013)
ET (mm/yr)	459 ^f	490	S, M	GLEAM (Martens et al., 2017)
Runoff (mm/yr)	286 ^f	297	G, M	UNH-GRDC (Fekete et al., 1999)

Note. The considered region is the European continent excluding Russia, Iceland, Cyprus, and Malta, but including Kaliningrad region and European part of Turkey, resulting in 3124 grid cells for our simulation. Values are representative only for the forest area of this region, except for tree cover, ET, and runoff, which are averaged over the entire region.

^aData source meanings: S, satellite-derived; I, inventory data; C, country reporting; M, modeled; G, Gauge-observations; F, flux measurements from ecological site; E, ecological site studies. ^bUsing aboveground biomass = 79% of vegetation biomass. ^cIncludes Turkey. ^dOnly upper 100 cm of soil considered in their study. Our value is for the upper 150cm. ^eTree cover here refers to canopy closure for all vegetation taller than 5 m (GFC) and LPJ-GUESS simulated crown cover of all simulated forests. GFC contains tree cover also for non-forest areas. ^fAssuming C3 grass growing everywhere outside of forest areas.

The values for simulated total harvest increased with higher-emission RCPs. The toNe management option showed higher harvests $(2.9-3.5 \text{ m}^3 \text{ha}^{-1} \text{ yr}^{-1})$ than toBd $(2.1-3.3 \text{ m}^3 \text{ha}^{-1} \text{ yr}^{-1})$, Figure S20 in Supporting Information S1), whereas converting to coppice caused strong reductions in harvests. Harvests for long-lived products (HLP) were much higher in the toNe management option than in toBd and lowest in coppice forests (Figure S21 in Supporting Information S1). For the unmanaged option harvests converged to zero by the end of the simulation period, as stands were successively left untouched.

Climate mitigation strongly depended on the RCP. For RCP2.6, the unmanaged forest had a higher cumulative mitigation potential than the base management (1.6 vs. 0.8 kgC/m^2), the situation was opposite for RCP8.5 (1.9 vs. 2.2 kgC/m^2).

The broad-leaved deciduous forests showed the highest roughness values, among other reasons due to the low plant area index in winter after senescence and a higher forest density than the needle-leaved evergreen forests. Also, the unmanaged forests showed higher surface roughness, majorly attributed to their tall trees.

Evapotranspiration was projected to be highest in coppices in our simulations (505 mm yr⁻¹ on average) but showed higher variation between the RCPs (Figure S5 in Supporting Information S1). Additionally, the management options revealed small differences in ET when averaging over the continent, with much larger differences locally (not shown).

The simulations also showed that soil water potential varied between RCPs due to different projected patterns in rainfall and CO_2 concentrations (Figure S10, Tables S3–S6 in Supporting Information S1). Soil water potentials of the base management option were projected on average 10% lower than present-day values. Different vegetation and management affected the soil water potential, with the lowest values in coppice (around -2.1 MPa).

Higher biodiversity habitat provision was observed in the unmanaged stands compared to other forest types. It was also higher in the toBd option compared to toNe, which had higher values than coppice.

3.2. Optimized Management Portfolios

3.2.1. Optimized Portfolios and Species Shares in European Regions

The portfolios were optimized for the 2100–2130 period but it is important to keep in mind that the different management transformations occurred gradually (Figure S22 in Supporting Information S1). Across Europe we



Southern Sweden (13.75, 55.75)







Figure 2. Example of the multi-criteria robust optimization approach for one grid cell. (a) Each box plot shows the spread of the normalized performance of the ecosystem service indicators (ESIs) over the four Representative Concentration Pathways (RCPs) when converting the forest of the entire grid cell to broad-leaved deciduous (toBd, green), to coppice (toCoppice, yellow), or to unmanaged forest (purple). ESIs for the other management options are omitted for clarity. Panel (b) is similar to (a), but for the optimized portfolio. Panel (c) shows the distribution of management options in the optimized portfolio for this grid cell. The figure illustrates how balanced the optimized portfolio is (b): regardless of the RCP, every ESI will be provided at a reasonable level (i.e., no ES performance will deteriorate in the future), whereas any other single management options. For example, in (a), in this grid cell, ET is highest for coppice for all RCPs, which is why the box plot is reduced to a single point at 1. This is not the case for, for example, unmanaged forest regarding mitigation: The box plots spans from 0.1 to 1 meaning that for some RCPs, mitigation is best for unmanaged, but for other RCPs it is quite bad (though never the worst, since the box does not go down to 0).

generally found that the optimized portfolios were quite diversified, comprised a large area of newly unmanaged forests, net increases in broad-leaved species shares, relevance of broad-leaved evergreen taxa in the South, and a divided importance of coppice (Figure 3).

Also, continually leaving a high share of the currently managed forest unattended played an important role in the optimized portfolios to maintain balanced ESIs. This share, 29% (Figure 3a), was similarly high across the different regions in Europe.

According to our optimization, the optimized species distribution generally shifted toward more broad-leaved species, especially in Northern Europe (Figure 3d) where the current high share of conifers decreased from 86% to 51%. Transitions in species proportion were visible in all regions with a net shift to more broad-leaved taxa from 38% to 55% by 2100–2130 across Europe (Figure 3a). However, for some grid cells the optimization caused a shift from broad-leaved species to needle-leaved species corresponding to 11% of current broad-leaved forests (mainly in Southern Europe). In comparison, roughly 34% of current needle-leaved forests were converted to broad-leaved ones by the next century (Figure 4). This transition included an increased extent of broad-leaved evergreen species (represented in this simulation by *Quercus ilex*) in Southern Europe with 22% of forests in that region consisting of broad-leaved evergreens (in the toBe, base, and unmanaged fractions of the portfolios).

According to our optimization, large regional differences were observed for coppice in the future forest management with no forests in the Southern and Alpine regions managed as coppice. At the same time, it had higher importance in Atlantic, Northern, and Continental grid cells (7%, 14%, and 7%, respectively, Figures 3c–3e).

3.2.2. Optimized Portfolio Compared to Present-Day Values and Base Management

According to our optimized portfolios (row "optimized" in Table 4 and Tables S3–S6 in Supporting Information S1), total carbon stocks were 1%–5% above present-day levels by the beginning of the next century. On the other hand, maintaining the base management (present-day species composition) in all regions across Europe decreased stocks by 3%–7%. The base management resulted in many regions becoming carbon sources, particularly in Central Europe and Southern Scandinavia (see Figure 5b, for visualization of RCP 4.5). Instead, the optimized portfolio provided more carbon storage in almost all grid cells than the base scenario (Figure 5d).

The mitigation ESI was higher in the optimized portfolio than base management in RCPs 2.6 and 4.5 whereas it was about equal in RCPs 6.0 and 8.5. When considering additionality, this means that in the lower RCPs, the optimized portfolio offered a positive net impact on climate change mitigation compared to base management, whereas in higher RCPs, there was no net effect on global climate.





Figure 3. Proposed management portfolios in Europe, aggregated over all simulated grid cells (a) and for different regions (b–f). The five regions (indicated by different shades of gray) are based on the climatic zones of Metzger et al. (2005) but aggregated as in the IPCC AR5 (Kovats et al., 2015). The leftmost chart in each box always shows the optimized share of management options in the entire region. The middle and right charts show the current and future shares of needle-leaved and broad-leaved tree species, respectively. The depiction of broad-leaved shares is shaded as it may contain both deciduous and evergreen species. Note that portfolios and future species shares differ between cells of a region. The computation is done for each grid cell independently (Figure S3 in Supporting Information S1). Here we show the forest-area-weighted aggregated shares.

The increase in carbon storage came at the cost of lower harvests compared to the base management option. Harvests for the optimized portfolio were simulated to be 4%-29% lower than present-day values, whereas for base they increased by 2%-33% depending on the RCP (Tables S3–S6 in Supporting Information S1). Additionally, Figure 6 shows the spatial patterns of these changes in the harvest volumes. Harvests were highest in Scandinavia, both today and in the future; however, in the optimized portfolio they decreased in various parts of Europe (Figure 6c).

Regarding the biogeophysical effects, roughness length was 17%–26% higher for the optimized portfolio than in present-day and 18%–21% higher than the future base values. These increases occurred in most regions except for a few cells (Figure S17c in Supporting Information S1). The optimized portfolio showed spatially different changes in ET with increases in Central, Eastern, and Northern Europe; however, they decreased in Atlantic and Southern regions (Figure S16c in Supporting Information S1). The effect of optimization regarding ET was





Figure 4. Net forest conversion from needle-leaved to broad-leaved forests in km². Negative values indicate a conversion from broad-leaved to needle-leaved forests.

small compared with the base management according to our simulations (Figure S16d in Supporting Information S1).

3.2.3. Optimized Portfolios With Alternative Assumptions

The previous results were based on the standard settings equally valuing all ESIs. However, decision-makers will likely encounter specific constraints or have regional preferences for some ESIs. Therefore, we briefly express our results when adding weights or constraints to the optimization.

First we found that counteracting the decrease in harvests by adding a constraint to keep total harvest at present-day levels in every grid cell is an unsuitable strategy: In about a fourth of the analyzed grid cells, such harvest levels could not be maintained until 2100–2130 for all RCPs since many stands did not reach the density and size conditions for a clear-cut. This made the optimization infeasible for many grid cells as the harvest constraint could not be met.

However, when valuing harvests twice as important as all other ESIs ("double harv"), the unmanaged fraction of the optimized portfolio was reduced to

18% (Figure S11 in Supporting Information S1). Harvests were higher than in the default optimization, ranging around the present-day levels when aggregating over the continent, nonetheless, they depended highly on the RCP (-16% to +12%, Table 4 and Tables S3–S6 in Supporting Information S1), with lower values for HLP.

Doubling the importance of climate mitigation ("double mit") resulted in a higher share of unmanaged forest such that 41% of currently managed European forests were proposed to be successively left untouched (Figure S12 in Supporting Information S1). This resulted in carbon stocks 5%–9% higher than present-day values, with a 4% higher mitigation potential than the standard optimization. There were also benefits for other ESIs compared with the standard optimization, but with even lower harvests than in the standard optimization portfolio. Additionally, coppice virtually vanished from the continent. Adding summer and winter albedo as ESIs did not change the overall portfolios much (Figure S14 in Supporting Information S1).

Furthermore, optimizing for RCP2.6 only led to slightly larger fractions of unmanaged forest (31% vs. 29%) and lower shares of coppice forests (3% vs. 7%). Also the total share of broad-leaved forests decreased from 55% to 49%

Ecosystem Service Indicator Mean Performances if One Management Were Applied to the 181 Grid Cells								
	C-Pool	Mitigation	Harvests	HLP	ET	Ψsoil	zO	Bio
Unit	kgC/m ²	kgC/m ²	m³/ha/yr	m³/ha/yr	mm/yr	MPa	m	unitless
base	9.8(±0.2)	$1.7(\pm 0.6)$	2.99(±0.36)	0.77(±0.09)	495(±22)	$-1.55(\pm 0.10)$	0.53(±0.01)	0.47(±0.01)
toBd	10.1(±0.1)	1.7(±0.7)	$2.77(\pm 0.52)$	0.43(±0.07)	498(±17)	$-1.66(\pm 0.12)$	$0.75(\pm 0.03)$	$0.55(\pm 0.02)$
toBe ^a	3.0(±0.1)	0.5(±0.2)	$0.78(\pm 0.08)$	0.12(±0.01)	457(±47)	$-2.76(\pm 0.08)$	$0.41(\pm 0.02)$	$0.08(\pm 0.00)$
toCoppice	9.9(±0.1)	1.3(±0.5)	1.77(±0.35)	0.11(±0.01)	505(±22)	$-2.12(\pm 0.10)$	$0.60(\pm 0.03)$	$0.21(\pm 0.01)$
toNe	9.7(±0.2)	$1.7(\pm 0.6)$	3.20(±0.29)	0.98(±0.10)	490(±25)	$-1.44(\pm 0.07)$	$0.40(\pm 0.01)$	$0.42(\pm 0.02)$
unmanaged	12.2(±0.2)	$1.8(\pm 0.2)$	0.08(±0.01)	$0.02(\pm 0.00)$	492(±23)	$-1.53(\pm 0.07)$	$0.66(\pm 0.04)$	$0.77(\pm 0.04)$
Present	10.3	0.0	2.48	0.65	501	-1.41	0.52	_b
Optimized	10.7(±0.1)	1.7(±0.5)	2.10(±0.29)	0.44(±0.05)	498(±22)	$-1.64(\pm 0.09)$	$0.63(\pm 0.02)$	$0.57(\pm 0.01)$
Double harv	$10.4(\pm 0.2)$	1.7(±0.6)	2.47(±0.32)	$0.55(\pm 0.06)$	498(±22)	$-1.62(\pm 0.09)$	$0.62(\pm 0.02)$	0.54(±0.01)
Double mit	11.0(±0.1)	1.8(±0.5)	1.75(±0.26)	$0.34(\pm 0.04)$	495(±21)	$-1.58(\pm 0.09)$	$0.67(\pm 0.02)$	$0.64(\pm 0.00)$

Note. The values are the area-weighted means averaged over the 4 RCPs, values in brackets are standard deviation and indicate the spread over the RCPs. The lower rows of the table correspond to the optimized portfolio in the standard setting ("optimized") as well as for the experiments with different weights and constraints as described in the text.

^atoBe management was only feasible in few regions in Europe, the reported values for this case are only for those regions (n = 38). In the optimized portfolio of a grid cell, only management was considered that ensured forest cover by the end of the century. ^bComputation of the Bio indicator is not sensible for present day, as it is solely a relative value between management options.

Table 4





Figure 5. (a) Simulated present-day and (b-d) future carbon storage (above and below ground in kgC/m²) in European forests for RCP4.5. Applying the base management (keeping species shares at present-day proportions, see Table 1) mainly led to decreases in carbon stock across Europe (b), the optimized management portfolio led to stable or increased carbon stocks in most parts of Europe (c), particularly in the East and North. By 2100–2130, the optimized portfolio had higher carbon stocks throughout Europe compared to the base scenario (d).

(Figure S15 in Supporting Information S1). Keeping displacement factors constant resulted in lower fractions of unmanaged forests (17%; in the Northern and Atlantic region around 10%), but a stronger focus on broad-leaved trees and coppice, compared to the standard optimization (Figure S13 in Supporting Information S1).

4. Discussion

4.1. Evaluation of LPJ-GUESS Simulation Results

LPJ-GUESS was able to reproduce present-day forest vegetation characteristics (Table 3 and Figures S6–S8 in Supporting Information S1) on which we based our management scenarios. However, soil carbon (and consequently total carbon) in our simulations was larger than estimates. This could result from the fact that we did not account for forest management prior to 1871. Since wood was never removed from the forest, trees that died simply decomposed on-site (Lindeskog et al., 2021). However, the estimate of Pan et al. (2011) also only considered the first 100 cm of the soil layer. A crucial point in projections via DGVMs is the response to changing environmental conditions, including elevated CO_2 -concentrations. In our runs, the increase in GPP between 1900 and 2010 was around 30% as observations also suggest (Figure S1 in Supporting Information S1, see Campbell et al., 2017). Regarding historical changes in the land sink, LPJ-GUESS shows an intermediate response compared to other models used in the Global Carbon Budget (Figure S2 in Supporting Information S1, see Friedlingstein et al., 2020).

4.2. Species Composition of Future Forests

According to our simulations and optimizations, implementing climate-smart forestry and adapting European forests to be productive under uncertain future climate conditions may imply strong changes in their composition, as well as trade-offs that need to be addressed.

First of all, throughout the continent a shift toward more broad-leaved species is proposed by our optimization. This result aligns with the reduce-air-temperature-portfolio of Luyssaert et al. (2018). However, it is in stark contrast to their maximize-carbon-sink-portfolio, proposing higher shares of conifers in the future. Since they optimized single objectives, their maximization of carbon sink focused mostly on coniferous forests due to their overall higher volumes and avoided emissions from substitution effects. On the other hand, our optimization



Figure 6. (a) Simulated present-day and ([b-d] future harvest; in m³ ha⁻¹ yr⁻¹ dry biomass) in European forests for RCP4.5. Harvests decreased in most regions in Europe in the optimized portfolios (c) but mostly increased if present-day management was extended into the future (b), leading to lower harvests for the optimized scenario compared to base (d). The main reason for this is the high share of unmanaged forests throughout Europe in the optimized portfolios.





Figure 7. Climate change mitigation of forests highly depends on the Representative Concentration Pathway (RCP): In this grid cell in Finland (21.75, 61.75), continuing the base management led to higher mitigation in RCP 8.5 compared to successively leaving the forest unmanaged (same species). This was different in low-RCP scenarios where the substitution of carbon-intensive fuels and materials through the usage of wood products was assumed to abate rather quickly, making the carbon storage in forests and products the main driver for mitigation potential. The reason for there still being harvests in the unmanaged option is that the change to unmanaged forest only happened gradually, since only once each stand reached maturity a decision about future management was made. See also Figures S20 and S22 in Supporting Information S1. The dashed lines show the mitigation if constant discounting factors are assumed, where in all RCPs, wood products keep avoiding the same amount of emissions today and in the future.

considered multiple ESIs, dampening the importance of conifers due to their negative performance in terms of other ESIs. Additionally, we assumed a higher risk of disturbance for needle-leaved compared to broad-leaved species (Figure S25 in Supporting Information S1, Pugh et al., 2019; Seidl et al., 2017), another reason for the lower needle-leaved shares in our portfolios. Since multiple factors govern air temperature, the reduce-air-temperature portfolio of Luyssaert et al. (2018) implicitly optimized for more than one variable. Therefore, its similarity to ours, in terms of species composition, is not surprising.

Propositions for more broad-leaved forests in Europe have previously been made regarding lower summer temperatures (Schwaab et al., 2020) and lower fire risk (Astrup et al., 2018). Felton et al. (2010) indicated similar potential benefits for biodiversity as our approach, particularly for replacing of coniferous monocultures in Scandinavia. Nevertheless, it must be noted that in our study, those broad-leaved forests were modeled using stands of one or two species. An actual implementation should naturally include more diverse forests.

Second, our simulations suggested a large (but gradual) shift toward more unmanaged forest attributed to its benefits for biodiversity, water availability, and surface roughness (see, e.g., Figure 2). These benefits of unmanaged forests regarding multiple ESs were also found in an MCDM case-study by Diaz-Balteiro et al. (2017). Also, we found that the role of unmanaged forests was divided regarding global climate change mitigation, strongly dependent on the assumptions about the decarbonization of the construction sector (Section 4.3, Figure 7). Luyssaert et al. (2018) similarly showed a 30% fraction of unmanaged forest in their maximize-carbon-sink-portfolio (note that their values reflect the entire European forest whereas we considered only the currently managed part). On the other hand, their reduce-air-temperature-portfolio only contained 19% of unmanaged forest. This matches our optimization results when we kept displacement factors constant as they did (see Section 3.2.3 and Figure S13 in Supporting Information S1). This emphasizes the importance of a thorough analysis of wood substitution effects and the carbon-intensity of other sectors (more detail in Section 4.3).

Third, coppice, currently an abundant forest type in Southern Europe with areas well above one third of total forest area in some countries (Maganotti et al., 2018), is proposed to vanish from Southern European forests by our optimization. The optimization penalized its disadvantages such as the missing provision of long-lived wood products, low surface roughness and water availability (Figure 8), the latter agreeing with findings by Drake et al. (2009), see also Hartwich et al. (2014). This computed decrease in coppice forests due to ES provision complements an observed and proposed transition of coppice forests toward other management forms in some regions of Southern Europe, in part due to economical aspects (e.g., Stajic et al., 2009; Vacchiano et al., 2017). In other areas where coppice is currently less relevant (but potentially used to be), its role could increase as the high LAI of dense coppice forests leads to high transpiration rates and consequently local cooling.



Southern Finland (23.75, 61.75)

a) ESI Performance 2100-2130 (indiv. management) b) ESI Performance 2100-2130 (optimized portfolio)

c) Optimized Portfolio Shares

f) Optimized Portfolio Shares



Southern Germany (9.75, 49.75)

d) ESI Performance 2100-2130 (indiv. management) e) ESI Performance 2100-2130 (optimized portfolio)





Greece (21.75, 39.75)

g) ESI Performance 2100-2130 (indiv. management) h) ESI Performance 2100-2130 (optimized portfolio)





Figure 8. Examples of the optimization across a North-South gradient. Plots (a, d, and g) show the worst cases over the Representative Concentration Pathways (RCPs): a value of 1 in these radar charts thus indicates that this management had the best performance for this Ecosystem service indicator (ESI) in any given RCP. A value of 0 consequently does not necessarily mean that this management was always the worst for this ESI, it only means that this management had the worst performance in at least one RCP. The optimized portfolio of the grid cells contained multiple forest types (c, f, and i) because each type had its own advantages and disadvantages (a, d, and g). Coppice was mixed into the portfolio for the Finnish cell (c) but not into that of the cell in Germany (f) because in the latter, the merits of coppice (i.e., ET) were not high enough to outweigh its disadvantages ([d] compare to [a]). The grid cell in Greece was dominated by broad-leaved evergreen forests (i) because it had many advantages. It contained a much smaller fraction of unmanaged forests because other than Ψ_{soil} , they performed poorly compared to the other options (g). Panels (b, e, and h) also show how the proposed portfolio provided the ESIs in a very balanced fashion: In the worst cases, all ESIs were provided at a balanced level (compare Figure 2). Note that we used $\lambda = 0.2$, thus not only optimizing for the worst case, but partly also for the average ESI performance which explains some peculiarities, for example, the high share of base management in (i).

> Fourth, our simulations indicated a trend to more drought-tolerant species especially in Southern Europe attributed to the low availability of water (Figure S10 in Supporting Information S1). Also, in our simulations, drought-induced higher mortality was intensified by lower regeneration due to the additional warmer winter temperatures that led to decreased natural reproduction. Consequently, species that are more drought-tolerant



and do not rely on winter chilling such as *Q. ilex* and *P. halepensis* were indicated to dominate in those regions, a result that is supported by the literature (López-Tirado & Hidalgo, 2018).

4.3. Forests and Climate Change Mitigation

The European forest carbon sink has been stable in recent decades (Grassi et al., 2019; Pan et al., 2011) but changes to forest management strategies already have been proposed necessary to sustain it (Nabuurs et al., 2013). One means of sustaining or increasing the sink could be to increase the forest area which happened in the past decades (Forest Europe, 2020) and could continue to do so (Perpiña Castillo et al., 2021). This however is not part of our study as we focused solely on "forests remaining forests." According to our simulations, keeping the area of managed forest constant will transform the sink into a source (Table 4, Tables S3–S6 in Supporting Information S1); however, leaving some forests unmanaged could sustain a small sink until the beginning of the next century (1%–5% increase in carbon stocks including products).

Nevertheless, climate change mitigation is not only about carbon storage. Substitution of fossil fuels and carbon-intensive products plays another significant role. Although many studies previously found that substitution effects are crucial for mitigation highlighting the importance of managed forests, recent research has indicated that the benefits of substitutions will likely decrease, depending on the RCP (Brunet-Navarro et al., 2021; Harmon, 2019). This trend is caused by an increased share of renewables in the energy mix and technological improvements in the construction sector (see, e.g., IEA, 2020; Lehne & Preston, 2018). Therefore, the importance of harvests for our mitigation ESI decreased since we explicitly considered such decreasing substitution effects (Figure S23 in Supporting Information S1) and optimized for multiple RCPs. Hence the high proposed share of unmanaged forests. This share even increased when only looking at RCP2.6 where substitution effects diminished very quickly (Figure S15 in Supporting Information S1). The opposite was true when keeping substitution effects constant for all RCPs (Figure S13 in Supporting Information S1).

It is important to note that our proposed shift to more unmanaged forests was gradual, that is, forests were successively taken out of management. Consequently, our approach still strongly utilized the substitution effects in the shorter term when they are still important in low emission scenarios. The total benefit of management or non-management in terms of mitigation depended on the time frame and RCP; hence our portfolios offer a diversified mitigation strategy.

Additionally, our assessment is somewhat unbalanced: The production of steel and concrete combined account for 14% of global carbon emissions, and their improvements in carbon-intensity partly depend on breakthrough technologies (IEA, 2020; Lehne & Preston, 2018). Although we accounted for such improvements, we ignored potential innovations in wood usage. Many new technologies will likely facilitate higher amounts of wood usage in the construction sector, including strategies to use wood from broad-leaved trees and increase the building materials' carbon content (Churkina et al., 2020). Considering this, the increased usage of wood as construction material (as opposed to burning) and longer product lifetimes could drastically affect the mitigation potential. Future studies should focus on this change which is currently rarely explicitly assessed (e.g., Dugan et al., 2018; Smyth et al., 2014, 2018). Another aspect that is important but out of scope is the impact of different materials and energy sources on other environmental and social dimensions (see, e.g., Blankendaal et al., 2014; Hertwich et al., 2016).

4.4. Management and Disturbances

Future management must consider increases in disturbances (Seidl et al., 2014, 2017; Senf & Seidl, 2021a, 2021b). A common strategy is to diversify risks against these disturbances (Jandl et al., 2019; Mitchell, 2013), which to some extent is implicitly included in our propositions: Since we assessed the provision of multiple ESIs under increasing disturbances, our optimization automatically resulted in diverse forest compositions. Additionally, our portfolios contained an increased share of broad-leaved forests which are less prone to disturbances (Figure S25 in Supporting Information S1, Astrup et al., 2018; Pugh et al., 2019). Due to the uncertainties and complexity related to disturbances (e.g., Ahlström et al., 2015; Reyer et al., 2017; Seidl et al., 2011), we chose a rather simple representation of the change of disturbance regimes. While we used species-specific disturbance intervals, we refrained from making their increases species-specific, too, because research has shown that also species that were previously thought to be rather resilient have experienced heavy impacts of disturbances in recent years, for example, European beech in Central Europe (Buras et al., 2020; Schuldt et al., 2020).

The direct effect of management regarding disturbances was only partly contained in our study, namely through salvage logging in managed stands and because in managed stands, some trees are already harvested and thus the impact of a disturbance is lower (Figure S25 in Supporting Information S1). The impact of salvage logging on mitigation was rather small since disturbed wood also remains in the forest for many years before it decays back to the atmosphere (Suzuki et al., 2019), and because we assumed low fractions of salvaged wood to be used in products. However, an increased material usage of salvaged wood and measures to dampen disturbance impacts will foster the relevance of forest management for mitigation. Examples are harvesting trees that are more likely to be affected by disturbance, adapting the age structure (O'Hara & Ramage, 2013), or active fire suppression (Agee et al., 2000; Fernandes et al., 2013). This should be addressed in more detail in future studies.

4.5. Harvest Reductions

The large fractions of successively unmanaged forest in the portfolios naturally resulted in a gradual reduction in harvest volumes in favor of other ESIs. Future harvests of the optimized portfolios in low-emission RCPs were simulated to be 20%–29% lower than present-day values, but closer to present-day values in the high-emission RCPs, likely due to simulated CO₂-fertilization (Tables S3–S6 in Supporting Information S1). Furthermore, long-lived wood products provision would be further reduced in the optimized portfolio as it includes broad-leaved forests and coppice, which provide lower shares of wood being useable for those products (especially coppice, almost completely used for firewood).

From a purely financial perspective, a decrease in harvests for mitigating climate change is unremarkable. Previous reports indicated that financial incentives are probably necessary to implement forest management measures for mitigation (e.g., Khanal et al., 2017). However, a decrease in harvests in Europe will lead to increased wood imports, possibly stemming from unsustainable sources leading to dislocation of emissions and increased ecological pressure in other regions (Berlik et al., 2002; Mayer et al., 2005).

The increasing demand for wood in various industries (FAO, 2022; Nabuurs et al., 2007) and a proposed timber usage rise for construction (Churkina et al., 2020) combined with increasing floor areas per capita (e.g., Bierwirth & Thomas, 2015) will cause great pressure on forests in terms of wood production. Therefore, this stands in direct conflict with the various other ESs offered by forests. Countermeasures could be to redirect harvests for fuel toward long-lived products (Churkina et al., 2020), and societal changes in the amount of wood required per capita. Other means could be forest expansion, changes in harvest intensity, and inclusion of fast-growing species such as Douglas fir (Thomas et al., 2022). Sensible intensities of such measures and their impact on ESs could also be assessed by our framework.

4.6. Provision of Ecosystem Services and Trade-Offs

Our study was aimed to assess the possibility of achieving climate-smart forestry by combining mitigation, adaptation, and the provision of other ESs, under the uncertainty of future climate pathways. Above we mainly focused on forest adaptation (e.g., some species are no longer suitable in certain regions, see Section 4.2) and the issues arising for mitigation and harvests. But our portfolios can help enabling the continued provision of other ESs.

Roughness length was estimated to be about 22% higher than present-day values. It is mostly governed by plant area index and canopy height (Moene & van Dam, 2014; Raupach, 1994). Higher roughness decreases aerodynamic resistance and consequently increases heat fluxes. The effect of canopy height explains the high simulated roughness values of unmanaged forests. Also, the change to more deciduous species increased the average roughness, since due to senescence their winter plant area index was quite low leading to lower resistance (Figure S9 in Supporting Information S1).

Future ET values heavily depended on the RCP, especially in the high-emission scenario, where we found strong ET reductions in all management options (Table S6, Figure S5 in Supporting Information S1). With elevated atmospheric CO_2 concentrations, plants do not need to open their stomata as much, hence losing less water (Keenan et al., 2013). Consequently, trees are more drought-resistant but the local cooling reduces, an implicit trade-off between adaptation and local climate regulation. However, in low-emission RCPs, this effect was not large enough to compensate for increased precipitation and temperature, causing a net increase in ET. Note that

vegetation-climate feedbacks will likely amplify temperature changes through ET, increasing warming in Southern Europe, and reducing warming in Central Europe (Wramneby et al., 2010).

Although our simulations indicated positive effects of elevated CO_2 levels on productivity through higher water use efficiency, soil water potentials decreased until the beginning of the next century for all RCPs except RCP2.6 (Figure S10 in Supporting Information S1). Furthermore, our water availability indicator depicting the minimal monthly soil water potential decreased for almost all management options and RCPs compared with the present-day values, indicating higher drought risk in the future. Needle-leaved forest showed the highest water availability—for RCP 2.6 even slightly higher than today. In contrast, coppice showed the lowest, due to its high LAI and water usage.

In terms of biodiversity, our approach favors unmanaged forests over managed ones due to the higher abundance of different age classes, dead biomass, and large trees. This result is confirmed by Vuidot et al. (2011) who argue that the number of different habitats makes unmanaged forests preferable in terms of biodiversity. However, more important than the "management versus conservation antagonism" is the diversity of landscapes (e.g., Schall et al., 2021) which our portfolio-based approach implicitly offers. Additionally, making management mimic features of unmanaged forests and especially avoiding clear-cuts with species changes could further improve the suitability of managed forests in terms of biodiversity (Paillet et al., 2010; Vuidot et al., 2011). Although diversity is often evaluated at a within-stand level, concentrating on landscape-scales similar to this study can be a reasonable approach (Schall et al., 2018). Importantly, using biodiversity indicators such as those used in this study is encouraged, though still speculative (Davies et al., 2008) and insufficient biodiversity assessment. Also, simulations with more heterogeneous species are necessary, for a larger variety of species and microhabitats (Vuidot et al., 2011).

4.7. Different Experiments and Applicability of the Methodology

In a few experiments (Section 3.2.3) we showed the implications of changing the optimization focus. Interestingly, a stronger focus on mitigation resulted in a near-complete disappearance of coppice management from the continent (Figure S12 in Supporting Information S1) due to low material substitution and low carbon stocks. Its apparent benefit for fuel substitution is also decreased since other management options obtain similar fuel substitutions because large fractions of wood products can displace fossil fuels at the end of their lifetime (Knauf et al., 2015). In the considered time frame, the fuel substitution of coppice thus did not have a significant impact. We consequently suggest that more research is necessary to assess the currently rising interest in coppice due to its apparent potential for bioenergy (Maganotti et al., 2018). Note that we mean traditional coppice here, not short-rotation plantations.

Inclusion of albedo only changed portfolios marginally (Figure S14 in Supporting Information S1) because of the strong focus on worst cases. Since we already assessed seven ESIs, chances were high that the "crucial" ESI was already included. Nevertheless, in some cells, including albedo led to changes, often toward more unmanaged forest, partly because these were simulated to be less dense, that is, some highly reflective grass shone through (not shown).

Increasing the importance of harvests in the optimization disclosed that harvest levels can be increased compared with the standard optimization while still adequately representing other ESs (Section 4.5), with harvests spread around present-day values, depending on the RCP. Adding constraints to the optimization to keep harvests at present-day levels in every grid cell could not provide a mathematically feasible solution for many grid cells. This was because harvest rates in these grid cells inevitably decreased until the end of the century, demonstrating that collaboration between countries will be necessary to meet future wood demands and climate objectives.

Our weighting was done equally for the entire continent, but the importance of ESs actually depends on the geographic location and local socioeconomic circumstances. Furthermore, our weights were independent of the RCP although different RCPs could yield different ESs to be most important. While technically possible, including ES-weights that depend on the RCP is a major task of its own and was excluded from this study. We would like to note however, that adaptation was accounted for as RCP-dependent in our study, since Equation 6 guarantees a healthy forest cover under all RCPs.

Considering all RCPs simultaneously is a very risk-averse strategy, especially considering the decreasing likelihood of RCP8.5 (Hausfather & Peters, 2020). However, including the full range of uncertainty could avoid costly adaptation (Lawrence et al., 2020) and is also suggested by the IPCC (IPCC, 2021). We however also applied our methodology to RCP2.6 only where at least the shift to broad-leaved forests is not as pronounced as when considering all RCPs (Figure S15 in Supporting Information S1).

Finally, our results showed management options and potential impacts on a very coarse $0.5^{\circ} \times 0.5^{\circ}$ resolution. In practice, an application of our methodology must be made on a fine scale taking into account site-specific characteristics, ideally with more detailed regional assessments of the inputs. Our general results can serve as a guideline for such an application, in which also ES preferences should be included, obtained for example by conducting regional stakeholder surveys (see, e.g., Knoke et al., 2020). Some ESs could also be ignored by setting weights to zero. This could for example, be the case when the mitigation potential of a particular forest is rather small and other ESs are more important. Additionally, harvest adjacency and avoidance of fragmentation (Baskent & Keles, 2005; Millar et al., 2007) could be directly included as constraints in our optimization, but lead to increased (computational) complexity.

4.8. Conclusion

The goal of our study was to obtain insights on possible climate-smart forest management portfolios across Europe allowing for a balanced provision of many ESs. The resulting portfolios with high shares of broad-leaved forests and unmanaged forests revealed a consequent trade-off between better performance of a few ESs against a balanced provision of all ESs with potentially lower performances.

The main trade-off shown in this study lies in harvest reductions in favor of other ESs, such as increased biodiversity, mitigation, and local cooling. Such harvest reductions and potentially increasing wood demand will put additional pressure on forest ecosystems, especially when forest areas remain constant as assumed here. This might also entail further ecological issues when wood is imported from unsustainable sources. Possible countermeasures might be increasing the fraction of wood harvests used for long-lived products (Churkina et al., 2020) and changing societal behavior. We also indicated how adding a preference to wood harvests in the optimization could alleviate the harvesting decreases to some extent; however, at the cost of other ESs.

Furthermore, our approach revealed another trade-off related to climate uncertainty. Portfolios optimized for a various climatic futures show weaker performance in some ESs than optimizing for a certain future, a cost of deriving robust solutions. However, concurrently, this is an important benefit of our approach: the possibility to make valid propositions for a wide range of climatic futures. We found this uncertainty-related trade-off to be particularly pronounced when assessing climate change mitigation. Mitigation depends on the forest carbon sink as well as wood products that substitute carbon-intensive materials and fuels in various economic sectors. However, in low-emission scenarios, the potential future decarbonization of these sectors may cause a decreased importance of substitution effects compared to the carbon sink, whereas the opposite is true in high-emission scenarios. Our portfolios offer an intermediate mitigation strategy, based on both, enhancing the carbon sink and continued material and fuel substitution. In any case, a crucial means for enhancing mitigation is increasing the fraction of harvested wood used for long-lived products. Further work should focus on this aspect and more detailed assessments of management options and ESIs. Also, the uncertainty space should be further explored by including forcing data from multiple GCMs and multiple process formulations in other DGVMs.

Conclusively, we applied a combination of forest management simulations with a multi-criteria robust optimization framework. Our study gives recommendations on climate-smart forest management options capable of providing many ESs in the future under a broad range of future climate scenarios. Our results provide insights on a general direction of European climate-smart forestry that may be used as a baseline when developing regional forest management strategies in practice. However, we also revealed that such a holistic approach does not eliminate all trade-offs. Although we acknowledge that further considerations are required, this study lays the groundwork for future research that examines trade-offs of forest management strategies in terms of ES provisioning and especially climate change mitigation under a highly uncertain future.



Data Availability Statement

The code to compute the ESIs, analyze them, and run the optimization is publicly available at https://doi. org/10.5281/zenodo.6667489 together with data for the four grid cells used for the illustrations. The entire data set to reproduce all optimizations for the European continent is available at https://doi.org/10.5281/ zenodo.6612953.

References

- Agee, J. K., Bahro, B., Finney, M. A., Omi, P. N., Sapsis, D. B., Skinner, C. N., et al. (2000). The use of shaded fuelbreaks in landscape fire management. Forest Ecology and Management, 127(1–3), 55–66. https://doi.org/10.1016/S0378-1127(99)00116-4
- Ahlström, A., Schurgers, G., Arneth, A., & Smith, B. (2012). Robustness and uncertainty in terrestrial ecosystem carbon response to CMIP5 climate change projections. *Environmental Research Letters*, 7(4), 044008. https://doi.org/10.1088/1748-9326/7/4/044008
- Ahlström, A., Xia, J., Arneth, A., Luo, Y., & Smith, B. (2015). Importance of vegetation dynamics for future terrestrial carbon cycling. *Environmental Research Letters*, 10(5), 054019. https://doi.org/10.1088/1748-9326/10/5/054019
- Albert, K., & Ammer, C. (2012). Biomasseproduktivität ausgewählter europäischer Mittel- und Niederwaldbestände Ergebnisse einer vergleichenden Metaanalyse. Allgemeine Forst und Jagdzeitung, 183, 225–237.
- Arehart, J. H., Hart, J., Pomponi, F., & D'Amico, B. (2021). Carbon sequestration and storage in the built environment. Sustainable Production and Consumption, 27, 1047–1063. https://doi.org/10.1016/j.spc.2021.02.028
- Astrup, R., Bernier, P. Y., Genet, H., Lutz, D. A., & Bright, R. M. (2018). A sensible climate solution for the boreal forest. Nature Climate Change, 8(1), 11–12. https://doi.org/10.1038/s41558-017-0043-3
- Baskent, E. Z., & Keles, S. (2005). Spatial forest planning: A review. *Ecological Modelling*, 188(2–4), 145–173. https://doi.org/10.1016/j. ecolmodel.2005.01.059
- Bellassen, V., Le Maire, G., Dhôte, J. F., Ciais, P., & Viovy, N. (2010). Modelling forest management within a global vegetation model—Part 1: Model structure and general behaviour. *Ecological Modelling*, 221(20), 2458–2474. https://doi.org/10.1016/j.ecolmodel.2010.07.008
- Ben-Tal, A., & Nemirovski, A. (2002). Robust optimization—Methodology and applications. *Mathematical Programming*, 92(3), 453–480. https://doi.org/10.1007/s101070100286
- Berlik, M. M., Kittredge, D. B., & Foster, D. R. (2002). The illusion of preservation: A global environmental argument for the local production of natural resources. *Journal of Biogeography*, 29(10–11), 1557–1568. https://doi.org/10.1046/j.1365-2699.2002.00768.x
- Bierwirth, A., & Thomas, S. (2015). Almost best friends: Sufficiency and efficiency; can sufficiency maximise efficiency gains in buildings? ECEEE 2015 - European Council for an Energy Efficient Economy (pp. 71–82).
- Binder, S., Haight, R. G., Polasky, S., Warziniack, T., Mockrin, M. H., Deal, R. L., & Arthaud, G. (2017). Assessment and valuation of forest ecosystem services: State of the science review (Vol. 170). Retrieved from https://www.fs.fed.us/nrs/pubs/gtr/gtr_nrs170.pdf
- Blankendaal, T., Schuur, P., & Voordijk, H. (2014). Reducing the environmental impact of concrete and asphalt: A scenario approach. Journal of Cleaner Production, 66, 27–36. https://doi.org/10.1016/j.jclepro.2013.10.012
- Booth, M. S., Mackey, B., & Young, V. (2020). It's time to stop pretending burning forest biomass is carbon neutral. *GCB Bioenergy*, 12(12), 1036–1037. https://doi.org/10.1111/gcbb.12716
- Brockerhoff, E. G., Barbaro, L., Castagneyrol, B., Forrester, D. I., Gardiner, B., González-Olabarria, J. R., et al. (2017). Forest biodiversity, ecosystem functioning and the provision of ecosystem services. *Biodiversity & Conservation*, 26(13), 3005–3035. https://doi.org/10.1007/s10531-017-1453-2
- Brunet-Navarro, P., Jochheim, H., Cardellini, G., Richter, K., & Muys, B. (2021). Climate mitigation by energy and material substitution of wood products has an expiry date. *Journal of Cleaner Production*, 303, 127026. https://doi.org/10.1016/j.jclepro.2021.127026
- Brus, D. J., Hengeveld, G. M., Walvoort, D. J., Goedhart, P. W., Heidema, A. H., Nabuurs, G. J., & Gunia, K. (2012). Statistical mapping of tree species over Europe. European Journal of Forest Research, 131(1), 145–157. https://doi.org/10.1007/s10342-011-0513-5
- Buras, A., Rammig, A., & Zang, C. S. (2020). Quantifying impacts of the 2018 drought on European ecosystems in comparison to 2003. Biogeosciences, 17(6), 1655–1672. https://doi.org/10.5194/bg-17-1655-2020
- Campbell, J., Berry, J., Seibt, U., Smith, S. J., Montzka, S. A., Launois, T. et al. (2017). Large historical growth in global terrestrial gross primary production. *Nature* 544, 84–87. https://doi.org/10.1038/nature22030
- Canadell, J. G., & Raupach, M. R. (2008). Managing forests for climate change mitigation. *Science*, 320(5882), 1456–1457. https://doi.org/10.1126/science.1155458
- Chang, J., Ciais, P., Wang, X., Piao, S., Asrar, G., Betts, R., et al. (2017). Benchmarking carbon fluxes of the ISIMIP2a biome models. *Environmental Research Letters*, 12(4), 045002. https://doi.org/10.1088/1748-9326/aa63fa
- Cherubini, F., Peters, G. P., Berntsen, T., Strømman, A. H., & Hertwich, E. (2011). CO₂ emissions from biomass combustion for bioenergy: Atmospheric decay and contribution to global warming. *GCB Bioenergy*, 3(5), 413–426. https://doi.org/10.1111/j.1757-1707.2011.01102.x
- Churkina, G., Organschi, A., Reyer, C. P., Ruff, A., Vinke, K., Liu, Z., et al. (2020). Buildings as a global carbon sink. *Nature Sustainability*, 3(4), 269–276. https://doi.org/10.1038/s41893-019-0462-4
- Cordonnier, T., Berger, F., Elkin, C., Lämås, T., Martinez, M., Bugmann, H., et al. (2014). ARANGE D2.2 Models and linker functions (indicators) for ecosystem services. (289437).
- Dai, A. (2013). Increasing drought under global warming in observations and models. *Nature Climate Change*, 3(1), 52–58. https://doi.org/10.1038/nclimate1633
- Davies, Z. G., Tyler, C., Stewart, G. B., & Pullin, A. S. (2008). Are current management recommendations for saproxylic invertebrates effective? A systematic review. *Biodiversity & Conservation*, 17(1), 209–234. https://doi.org/10.1007/s10531-007-9242-y
- Diaz-Balteiro, L., Alonso, R., Martínez-Jaúregui, M., & Pardos, M. (2017). Selecting the best forest management alternative by aggregating ecosystem services indicators over time: A case study in central Spain. *Ecological Indicators*, 72, 322–329. https://doi.org/10.1016/j.ecolind.2016.06.025
- Diaz-Balteiro, L., Belavenutti, P., Ezquerro, M., González-Pachón, J., Ribeiro Nobre, S., & Romero, C. (2018). Measuring the sustainability of a natural system by using multi-criteria distance function methods: Some critical issues. *Journal of Environmental Management*, 214, 197–203. https://doi.org/10.1016/j.jenvman.2018.03.005
- Drake, P. L., Mendham, D. S., White, D. A., & Ogden, G. N. (2009). A comparison of growth, photosynthetic capacity and water stress in *Eucalyptus globulus* coppice regrowth and seedlings during early development. *Tree Physiology*, 29(5), 663–674. https://doi.org/10.1093/treephys/tpp006

Acknowledgments

The authors gratefully acknowledge the computational and data resources provided by the Leibniz Supercomputing Centre (www.lrz.de). We would also like to thank Heinz-Detlef Gregor, Wolfgang Falk, and Thomas Hickler for fruitful discussions. We also acknowledge funding from the ERA-Net "Sumforest" project "FOREXCLIM" (Grant No. 2816ERA01S), and the Bavarian State Ministry of Science and the Arts in the context of the Bavarian Climate Research Network (bayklif) through its BLIZ project (Grant No. 7831-26625-2017, www.bayklif-bliz.de). We further acknowledge funding from the European Forest Institute (EFI) Networking Fund FORMASAM. Open Access funding enabled and organized by Projekt DEAL.

- Dufresne, J.-L., Foujols, M.-A., Denvil, S., Caubel, A., Marti, O., Aumont, O., et al. (2013). Climate change projections using the IPSL-CM5 Earth System Model: From CMIP3 to CMIP5. *Climate Dynamics*, 40(9–10), 2123–2165. https://doi.org/10.1007/s00382-012-1636-1
- Dugan, A. J., Birdsey, R., Mascorro, V. S., Magnan, M., Smyth, C. E., Olguin, M., & Kurz, W. A. (2018). A systems approach to assess climate change mitigation options in landscapes of the United States forest sector. *Carbon Balance and Management*, 13(1), 13. https://doi. org/10.1186/s13021-018-0100-x
- Díaz, S., Fargione, J., Chapin, F. S., & Tilman, D. (2006). Biodiversity loss threatens human well-being. PLoS Biology, 4(8), 1300–1305. https:// doi.org/10.1371/journal.pbio.0040277

Eurostat. (2021a). Eurostat forestry database. Retrieved from https://ec.europa.eu/eurostat/web/forestry/data/database

Eurostat. (2021b). Landfill rate of waste excluding major mineral wastes. Retrieved from https://ec.europa.eu/eurostat/databrowser/view/ t2020_rt110/default/table?lang=en

Evans, J. (1984). Silviculture of broadleaved woodland. Forestry Commission Bulletin, 62.

- FAO. (2020). Food and Agriculture Organization of the United Nations: Global forest resources assessment 2020: Terms and definition FRA. Retrieved from http://www.fao.org/forestry/58864/en/
- FAO. (2022). FAOSTAT database. Forestry production and trade. Retrieved from https://www.fao.org/faostat/en/%23data/FO
 - Fekete, B. M., Vörösmarty, C. J., & Grabs, W. (1999). Global, composite runoff fields based on observed river discharge and simulated water balances. Global Runoff Data Centre Koblenz.
 - Felton, A., Lindbladh, M., Brunet, J., & Fritz, Ö. (2010). Replacing coniferous monocultures with mixed-species production stands: An assessment of the potential benefits for forest biodiversity in northern Europe. *Forest Ecology and Management*, 260(6), 939–947. https://doi.org/10.1016/j.foreco.2010.06.011
 - Fernandes, P. M., Davies, G. M., Ascoli, D., Fernández, C., Moreira, F., Rigolot, E., et al. (2013). Prescribed burning in southern Europe: Developing fire management in a dynamic landscape. Frontiers in Ecology and the Environment, 11(S1). https://doi.org/10.1890/120298
 - Forest Europe. (2015a). Seventh Ministerial Conference on the Protection of Forest in Europe (pp. 1–8). Retrieved from https://foresteurope. org/wp%2Dcontent/uploads/2016/11/III.%2DELM%5F7MC%5F2%5F2015%5FMinisterialDeclaration%5Fadopted%2D2.pdf%23page%3D5 Forest Europe. (2015b). State of Europe's forests 2015.
 - Forest Europe. (2020). State of Europe's forests 2020.
 - Friedlingstein, P., O'Sullivan, M., Jones, M. W., Andrew, R. M., Hauck, J., Olsen, A., et al. (2020). Global carbon budget 2020. Earth System Science Data, 12(4), 3269–3340. https://doi.org/10.5194/essd-12-3269-2020
 - Gorissen, B. L., Yanikoğlu, I., & den Hertog, D. (2015). A practical guide to robust optimization. Omega, 53, 124–137. https://doi.org/10.1016/j. omega.2014.12.006
 - Grassi, G., Cescatti, A., Matthews, R., Duveiller, G., Camia, A., Federici, S., et al. (2019). On the realistic contribution of European forests to reach climate objectives. *Carbon Balance and Management*, 14(1), 1–5. https://doi.org/10.1186/s13021-019-0123-y
 - Grassi, G., Fiorese, G., Pilli, R., Jonsson, K., Blujdea, V., Korosuo, A., & Vizzarri, M. (2021). Brief on the role of the forest-based bioeconomy in mitigating climate change through carbon storage and material substitution (pp. 1–16). Retrieved from https://publications.jrc.ec.europa.eu/ repository/handle/JRC124374
 - Hansen, M. C., Potapov, P. V., Moore, R., Hancher, M., Turubanova, S. A., Tyukavina, A., et al. (2013). High-resolution global maps of 21st-century forest cover change. *Science*, 342(6160), 850–853. https://doi.org/10.1126/science.1244693
 - Harmon, M. E. (2019). Have product substitution carbon benefits been overestimated? A sensitivity analysis of key assumptions. *Environmental Research Letters*, 14(6), 065008. https://doi.org/10.1088/1748-9326/ab1e95
 - Harris, N. L., Gibbs, D. A., Baccini, A., Birdsey, R. A., de Bruin, S., Farina, M., et al. (2021). Global maps of twenty-first century forest carbon fluxes. *Nature Climate Change*, 11(3), 234–240. https://doi.org/10.1038/s41558-020-00976-6
 - Hartwich, J., Bölscher, J., & Schulte, A. (2014). Impact of short-rotation coppice on water and land resources. *Water International*, 39(6), 813–825. https://doi.org/10.1080/02508060.2014.959870
 - Hausfather, Z., & Peters, G. P. (2020). Emissions The 'business as usual' story is misleading. *Nature*, 577(7792), 618–620. https://doi.org/10.1038/d41586-020-00177-3
 - Haverd, V., Smith, B., Canadell, J. G., Cuntz, M., Mikaloff-Fletcher, S., Farquhar, G., et al. (2020). Higher than expected CO₂ fertilization inferred from leaf to global observations. *Global Change Biology*, 26(4), 2390–2402. https://doi.org/10.1111/gcb.14950
 - Haxeltine, A., & Prentice, I. C. (1996). BIOME3: An equilibrium terrestrial biosphere model based on ecophysiological constraints, resource availability, and competition among plant functional types. *Global Biogeochemical Cycles*, 10(4), 693–709. https://doi.org/10.1029/96GB02344
 - Hertwich, E., de Larderel, J. A., Arvesen, A., Bayer, P., Bergesen, J., Bouman, E., et al. (2016). Green Energy Choices: The benefits, risks, and trade-offs of low-carbon technologies for electricity production (Technical Report). UNEP.
 - Hickler, T., Vohland, K., Feehan, J., Miller, P. A., Smith, B., Costa, L., et al. (2012). Projecting the future distribution of European potential natural vegetation zones with a generalized, tree species-based dynamic vegetation model. *Global Ecology and Biogeography*, 21(1), 50–63. https://doi.org/10.1111/j.1466-8238.2010.00613.x
 - Holtsmark, B. (2012). Harvesting in boreal forests and the biofuel carbon debt. Climatic Change, 112(2), 415–428. https://doi.org/10.1007/s10584-011-0222-6
 - Howard, C., Dymond, C. C., Griess, V. C., Tolkien-Spurr, D., & van Kooten, G. C. (2021). Wood product carbon substitution benefits: A critical review of assumptions. *Carbon Balance and Management*, *16*(1), 1–11. https://doi.org/10.1186/s13021-021-00171-w
 - Hua, F., Bruijnzeel, L. A., Meli, P., Martin, P. A., Zhang, J., Nakagawa, S., et al. (2022). The biodiversity and ecosystem service contributions and trade-offs of forest restoration approaches. *Science*, 376(6595), 839–844. https://doi.org/10.1126/science.abl4649
 - IEA. (2020). Iron and steel technology roadmap (Technical Report). Retrieved from https://www.iea.org/reports/iron-and-steel-technology-roadmap (Technical Report). Technical Report (Technical Report) and the steel technology roadmap (Technical Report). Technical Report (Technical Report) and the steel technology roadmap (Technical Report). Technical Report (Technical Report) and the steel technology roadmap (Technical Report). Technical Report (Technical Report) and the steel technology roadmap (Technical Report). Technical Report (Technical Report) and the steel technology roadmap (Technical Report). Technical Report (Technical Report) and the steel technology roadmap (Technical Report). Technical Report (Technical Report) and the steel technology roadmap (Technical Report). Technical Report (Technical Report) and technical Report (Technical Report). Technical Report (Technical Report) and technical Report (Technical Report). Technical Report (Technical Report) and technical Report (Technical Report) and technical Report (Technical Report). Technical Report (Technical Report) and technical Report (Technical Report) and technical Report (Technical Report). Technical Report (Technical Report) and technical Report (Technical R
 - IPCC. (2014). Summary for policymakers. In Climate change 2013—The physical science basis: Working Group I Contribution to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 1–30). Cambridge University Press. https://doi.org/10.1017/ CBO9781107415324.004
 - IPCC. (2018). Proposed outline of the special report in 2018 on the impacts of global warming of 1.5°C above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change. IPCC - Sr15, 2, 17–20.
 - IPCC. (2021). Summary for policymakers. In Climate change 2021—The physical science basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change.
 - Ishizaka, A., & Nemery, P. (2013). Multi-criteria decision analysis. John Wiley & Sons Ltd. https://doi.org/10.1002/9781118644898
 - Ito, A., Nishina, K., Reyer, C. P. O., François, L., Henrot, A.-J., Munhoven, G., et al. (2017). Photosynthetic productivity and its efficiencies in ISIMIP2a biome models: Benchmarking for impact assessment studies. *Environmental Research Letters*, 12(8), 085001. https://doi.org/10.1088/1748-9326/aa7a19

- Jandl, R., Spathelf, P., Bolte, A., & Prescott, C. E. (2019). Forest adaptation to climate change—Is non-management an option? Annals of Forest Science, 76(2), 1–13. https://doi.org/10.1007/s13595-019-0827-x
- Johansson, T. (2008). Sprouting ability and biomass production of downy and silver birch stumps of different diameters. *Biomass and Bioenergy*, 32(10), 944–951. https://doi.org/10.1016/j.biombioe.2008.01.009
- Kauppi, P., Hanewinkel, M., Lundmark, T., Nabuurs, G.-J., Peltola, H., Trasobares, A., & Hetemäki, L. (2018). Climate smart forestry in Europe. European Forest Institute.
- Keenan, T. F., Hollinger, D. Y., Bohrer, G., Dragoni, D., Munger, J. W., Schmid, H. P., & Richardson, A. D. (2013). Increase in forest water-use efficiency as atmospheric carbon dioxide concentrations rise. *Nature*, 499(7458), 324–327. https://doi.org/10.1038/nature12291
- Khanal, P. N., Grebner, D. L., Munn, I. A., Grado, S. C., Grala, R. K., & Henderson, J. E. (2017). Evaluating non-industrial private forest landowner willingness to manage for forest carbon sequestration in the southern United States. *Forest Policy and Economics*, 75, 112–119. https:// doi.org/10.1016/j.forpol.2016.07.004
- Klein, D., Höllerl, S., Blaschke, M., & Schulz, C. (2013). The contribution of managed and unmanaged forests to climate change mitigation—A model approach at stand level for the main tree species in Bavaria. *Forests*, 4(1), 43–69. https://doi.org/10.3390/f4010043
- Knauf, M., Joosten, R., & Frühwald, A. (2016). Assessing fossil fuel substitution through wood use based on long-term simulations. Carbon Management, 7(1–2), 67–77. https://doi.org/10.1080/17583004.2016.1166427
- Knauf, M., Köhl, M., Mues, V., Olschofsky, K., & Frühwald, A. (2015). Modeling the CO₂-effects of forest management and wood usage on a regional basis. *Carbon Balance and Management*, 10(1), 1–12. https://doi.org/10.1186/s13021-015-0024-7
- Knoke, T., Paul, C., Hildebrandt, P., Calvas, B., Castro, L. M., Hartl, F., et al. (2016). Compositional diversity of rehabilitated tropical lands supports multiple ecosystem services and buffers uncertainties. *Nature Communications*, 7(1), 11877. https://doi.org/10.1038/ncomms11877
- Knoke, T., Paul, C., Härtl, F., Castro, L. M., Calvas, B., & Hildebrandt, P. (2015). Optimizing agricultural land-use portfolios with scarce data—A non-stochastic model. *Ecological Economics*, 120, 250–259. https://doi.org/10.1016/j.ecolecon.2015.10.021
- Knoke, T., Paul, C., Rammig, A., Gosling, E., Hildebrandt, P., Härtl, F., et al. (2020). Accounting for multiple ecosystem services in a simulation of land-use decisions: Does it reduce tropical deforestation? *Global Change Biology*, *26*(4), 2403–2420. https://doi.org/10.1111/gcb.15003
- Kovats, R. S., Valentini, R., Bouwer, L. M., Georgopoulou, E., Jacob, D., Martin, E., et al. (2015). Europe. In V. R. Barros, C. B. Field, D. J. Dokken, M. D. Mastrandrea, & K. J. Mach (Eds.), *Climate change 2014: Impacts, adaptation and vulnerability* (pp. 1267–1326). Cambridge University Press. https://doi.org/10.1017/CBO9781107415386.003
- Krause, A., Knoke, T., & Rammig, A. (2020). A regional assessment of land-based carbon mitigation potentials: Bioenergy, BECCS, reforestation, and forest management. GCB Bioenergy, 12(5), 346–360. https://doi.org/10.1111/gcbb.12675
- Lamarque, J. F., Kyle, P. P., Meinshausen, M., Riahi, K., Smith, S. J., van Vuuren, D. P., et al. (2011). Global and regional evolution of short-lived radiatively-active gases and aerosols in the Representative Concentration Pathways. *Climatic Change*, 109(1), 191–212. https://doi. org/10.1007/s10584-011-0155-0
- Lawrence, J., Haasnoot, M., & Lempert, R. (2020). Climate change: Making decisions in the face of deep uncertainty. Nature, 580(7804), 456. https://doi.org/10.1038/d41586-020-01147-5
- Lehne, J., & Preston, F. (2018). Making concrete change; innovation in low-carbon Cement and concrete (Technical Report). Retrieved from www.chathamhouse.org
- Leonardsson, J., & Götmark, F. (2015). Differential survival and growth of stumps in 14 woody species after conservation thinning in mixed oak-rich temperate forests. *European Journal of Forest Research*, 134(1), 199–209. https://doi.org/10.1007/s10342-014-0843-1
- Leturcq, P. (2020). GHG displacement factors of harvested wood products: The myth of substitution. *Scientific Reports*, 10(1), 1–9. https://doi.org/10.1038/s41598-020-77527-8
- Lindeskog, M., Smith, B., Lagergren, F., Sycheva, E., Ficko, A., Pretzsch, H., & Rammig, A. (2021). Accounting for forest management in the estimation of forest carbon balance using the dynamic vegetation model LPJ-GUESS (v4.0, r9710): Implementation and evaluation of simulations for Europe. *Geoscientific Model Development*, 14(10), 6071–6112. https://doi.org/10.5194/gmd-14-6071-2021
- Lindner, M., Fitzgerald, J. B., Zimmermann, N. E., Reyer, C., Delzon, S., van der Maaten, E., et al. (2014). Climate change and European forests: What do we know, what are the uncertainties, and what are the implications for forest management? *Journal of Environmental Management*, 146, 69–83. https://doi.org/10.1016/j.jenvman.2014.07.030
- Liu, Y. Y., Van Dijk, A. I., De Jeu, R. A., Canadell, J. G., McCabe, M. F., Evans, J. P., & Wang, G. (2015). Recent reversal in loss of global terrestrial biomass. *Nature Climate Change*, 5(5), 470–474. https://doi.org/10.1038/nclimate2581
- Luyssaert, S., Ciais, P., Piao, S. L., Schulze, E.-d., Jung, M., Zaehle, S., et al. (2010). The European carbon balance. Part 3: Forests. *Global Change Biology*, 16(5), 1429–1450. https://doi.org/10.1111/j.1365-2486.2009.02056.x
- Luyssaert, S., Marie, G., Valade, A., Chen, Y.-Y., Djomo, S. N., Ryder, J., et al. (2018). Trade-offs in using European forests to meet climate objectives. *Nature*, 562(7726), 259–262. https://doi.org/10.1038/s41586-018-0577-1
- López-Tirado, J., & Hidalgo, P. J. (2018). Predicting suitability of forest dynamics to future climatic conditions: The likely dominance of Holm oak [Quercus ilex subsp. ballota (Desf.) Samp.] and Aleppo pine (Pinus halepensis Mill.). Annals of Forest Science, 75(1), 19. https://doi. org/10.1007/s13595-018-0702-1
- Maganotti, N., Schweier, J., Spinelli, R., Tolosana, E., Jylhä, P., Sopushynskyy, I., et al. (2018). Coppice forests in Europe.
- Marler, R. T., & Arora, J. S. (2004). Survey of multi-objective optimization methods for engineering. Structural and Multidisciplinary Optimization, 26(6), 369–395. https://doi.org/10.1007/s00158-003-0368-6
- Martens, B., Miralles, D. G., Lievens, H., van der Schalie, R., de Jeu, R. A. M., Fernández-Prieto, D., et al. (2017). GLEAM v3: Satellite-based land evaporation and root-zone soil moisture. *Geoscientific Model Development*, 10(5), 1903–1925. https://doi.org/10.5194/gmd-10-1903-2017
- Mauri, A., Strona, G., & San-Miguel-Ayanz, J. (2017). EU-Forest, a high-resolution tree occurrence dataset for Europe. *Scientific Data*, 4(1), 160123. https://doi.org/10.1038/sdata.2016.123
- Mayer, A. L., Kauppi, P. E., Angelstam, P. K., Zhang, Y., & Tikka, P. M. (2005). Importing timber, exporting ecological impact. Science, 308(5720), 359–360. https://doi.org/10.1126/science.1109476
- Meinshausen, M., Smith, S. J., Calvin, K., Daniel, J. S., Kainuma, M. L. T., Lamarque, J.-F., et al. (2011). The RCP greenhouse gas concentrations and their extensions from 1765 to 2300. *Climatic Change*, 109(1–2), 213–241. https://doi.org/10.1007/s10584-011-0156-z
- Metzger, M. J., Bunce, R. G. H., Jongman, R. H. G., Mücher, C. A., & Watkins, J. W. (2005). A climatic stratification of the environment of Europe. *Global Ecology and Biogeography*, 14(6), 549–563. https://doi.org/10.1111/j.1466-822X.2005.00190.x
- Millar, C. I., Stephenson, N. L., & Stephens, S. L. (2007). Climate change and forests of the future: Managing in the face of uncertainty. *Ecolog-ical Applications*, 17(8), 2145–2151. https://doi.org/10.1890/06-1715.1
- Millenium Ecosystem Assessment. (2005). Ecosystems and human well-being: Synthesis (Technical Report). Retrieved from https://www.millenniumassessment.org/documents/document.356.aspx.pdf

Mitchell, S. J. (2013). Wind as a natural disturbance agent in forests: A synthesis. Forestry, 86(2), 147–157. https://doi.org/10.1093/forestry/ cps058

Moene, A. F., & van Dam, J. C. (2014). Transport in the atmosphere-vegetation-soil continuum. Cambridge University Press. https://doi. org/10.1017/CBO9781139043137

Mori, A. S., Lertzman, K. P., & Gustafsson, L. (2017). Biodiversity and ecosystem services in forest ecosystems: A research agenda for applied forest ecology. *Journal of Applied Ecology*, 54(1), 12–27. https://doi.org/10.1111/1365-2664.12669

Myllyviita, T., Soimakallio, S., Judl, J., & Seppälä, J. (2021). Wood substitution potential in greenhouse gas emission reduction–review on current state and application of displacement factors. *Forest Ecosystems*, 8(1), 42. https://doi.org/10.1186/s40663-021-00326-8

Nabuurs, G. J., Delacote, P., Ellison, D., Hanewinkel, M., Hetemäki, L., Lindner, M., & Ollikainen, M. (2017). By 2050 the mitigation effects of EU forests could nearly double through climate smart forestry. *Forests*, 8(12), 1–14. https://doi.org/10.3390/f8120484

Nabuurs, G. J., Lindner, M., Verkerk, P. J., Gunia, K., Deda, P., Michalak, R., & Grassi, G. (2013). First signs of carbon sink saturation in European forest biomass. *Nature Climate Change*, 3(9), 792–796. https://doi.org/10.1038/nclimate1853

- Nabuurs, G. J., Pussinen, A., van Brusselen, J., & Schelhaas, M. J. (2007). Future harvesting pressure on European forests. European Journal of Forest Research, 126(3), 391–400. https://doi.org/10.1007/s10342-006-0158-y
- Nicolescu, V.-N., Carvalho, J., Hochbichler, E., Bruckman, V., Piqué-nicolau, M., Hernea, C., et al. (2017). Silvicultural guidelines for European coppice forests (Technical Report). Albert Ludwig University of Freiburg.
- O'Hara, K. L., & Ramage, B. S. (2013). Silviculture in an uncertain world: Utilizing multi-aged management systems to integrate disturbance. Forestry, 86(4), 401–410. https://doi.org/10.1093/forestry/cpt012
- Paillet, Y., Bergès, L., Hjältén, J., Ódor, P., Avon, C., Bernhardt-Römermann, M., et al. (2010). Biodiversity differences between managed and unmanaged forests: Meta-analysis of species richness in Europe. *Conservation Biology*, 24(1), 101–112. https://doi. org/10.1111/j.1523-1739.2009.01399.x

Pan, Y., Birdsey, R. A., Fang, J., Houghton, R., Kauppi, P. E., Kurz, W. A., et al. (2011). A large and persistent carbon sink in the world's forests. Science, 333(6045), 988–993. https://doi.org/10.1126/science.1201609

- Pedersen, J. S. T., van Vuuren, D. P., Aparício, B. A., Swart, R., Gupta, J., & Santos, F. D. (2020). Variability in historical emissions trends suggests a need for a wide range of global scenarios and regional analyses. *Communications Earth & Environment*, 1(1), 1–7. https://doi. org/10.1038/s43247-020-00045-y
- Perpiña Castillo, C., Jacobs-Crisioni, C., Diogo, V., & Lavalle, C. (2021). Modelling agricultural land abandonment in a fine spatial resolution multi-level land-use model: An application for the EU. *Environmental Modelling & Software*, 136, 104946. https://doi.org/10.1016/j. envsoft.2020.104946
- Pongratz, J., Reick, C. H., Raddatz, T., & Claussen, M. (2010). Biogeophysical versus biogeochemical climate response to historical anthropogenic land cover change. *Geophysical Research Letters*, 37(8), 1–5. https://doi.org/10.1029/2010GL043010
- Poulter, B., Aragão, L., Andela, N., Bellassen, V., Ciais, P., Kato, T., et al. (2018). The global forest age dataset (GFADv1.0) [data set]. PANGAEA. https://doi.org/10.1594/PANGAEA.889943
- Pugh, T. A. M., Arneth, A., Kautz, M., Poulter, B., & Smith, B. (2019). Important role of forest disturbances in the global biomass turnover and carbon sinks. *Nature Geoscience*, 12(9), 730–735. https://doi.org/10.1038/s41561-019-0427-2
- Rajasekaran, E., Das, N., Poulsen, C., Behrangi, A., Swigart, J., Svoboda, M., et al. (2018). SMAP soil moisture change as an indicator of drought conditions. *Remote Sensing*, 10(5), 788. https://doi.org/10.3390/rs10050788
- Ramstein, C., Dominioni, G., Ettehad, S., Lam, L., Quant, M., Zhang, J., et al. (2019). State and trends of carbon pricing 2019. The World Bank. https://doi.org/10.1596/978-1-4648-1435-8
- Raupach, M. R. (1994). Simplified expressions for vegetation roughness length and zero-plane displacement as functions of canopy height and area index. Boundary-Layer Meteorology, 71(1–2), 211–216. https://doi.org/10.1007/BF00709229
- Reineke, L. H. (1933). Perfecting a stand-density index for even-aged forest. Journal of Agricultural Research, 46, 627–638.
- Reyer, C. P., Bathgate, S., Blennow, K., Borges, J. G., Bugmann, H., Delzon, S., et al. (2017). Are forest disturbances amplifying or canceling out climate change-induced productivity changes in European forests? *Environmental Research Letters*, 12(3), 034027. https://doi. org/10.1088/1748-9326/aa5ef1
- Ruiz, F., Cabello, J. M., & Luque, M. (2011). An application of reference point techniques to the calculation of synthetic sustainability indicators. Journal of the Operational Research Society, 62(1), 189–197. https://doi.org/10.1057/jors.2009.187
- Rydberg, D. (2000). Initial sprouting, growth and mortality of European aspen and birch after selective coppicing in central Sweden. Forest Ecology and Management, 130(1-3), 27–35. https://doi.org/10.1016/S0378-1127(99)00187-5
- Rüter, S. (2011). Welchen Beitrag leisten Holzprodukte zur CO2-Bilanz? AFZ/Der Wald, 15, 15-18.
- Santangeli, A., Toivonen, T., Pouzols, F. M., Pogson, M., Hastings, A., Smith, P., & Moilanen, A. (2016). Global change synergies and trade-offs between renewable energy and biodiversity. GCB Bioenergy, 8(5), 941–951. https://doi.org/10.1111/gcbb.12299
- Sathre, R., & O'Connor, J. (2010). Meta-analysis of greenhouse gas displacement factors of wood product substitution. *Environmental Science & Policy*, 13(2), 104–114. https://doi.org/10.1016/j.envsci.2009.12.005
- Schall, P., Gossner, M. M., Heinrichs, S., Fischer, M., Boch, S., Prati, D., et al. (2018). The impact of even-aged and uneven-aged forest management on regional biodiversity of multiple taxa in European beech forests. *Journal of Applied Ecology*, 55(1), 267–278. https://doi. org/10.1111/1365-2664.12950
- Schall, P., Heinrichs, S., Ammer, C., Ayasse, M., Boch, S., Buscot, F., et al. (2021). Among stand heterogeneity is key for biodiversity in managed beech forests but does not question the value of unmanaged forests: Response to Bruun and Heilmann-Clausen (2021). Journal of Applied Ecology, 58(9), 1817–1826. https://doi.org/10.1111/1365-2664.13959
- Schuldt, B., Buras, A., Arend, M., Vitasse, Y., Beierkuhnlein, C., Damm, A., et al. (2020). A first assessment of the impact of the extreme 2018 summer drought on Central European forests. *Basic and Applied Ecology*, 45, 86–103. https://doi.org/10.1016/j.baae.2020.04.003
- Schwaab, J., Davin, E. L., Bebi, P., Duguay-Tetzlaff, A., Waser, L. T., Haeni, M., & Meier, R. (2020). Increasing the broad-leaved tree fraction in European forests mitigates hot temperature extremes. *Scientific Reports*, 10(1), 1–9. https://doi.org/10.1038/s41598-020-71055-1

Seidl, R., Rammer, W., & Lexer, M. J. (2011). Adaptation options to reduce climate change vulnerability of sustainable forest management in the Austrian Alps. Canadian Journal of Forest Research, 41(4), 694–706. https://doi.org/10.1139/x10-235

- Seidl, R., Schelhaas, M.-J., Rammer, W., & Verkerk, P. J. (2014). Increasing forest disturbances in Europe and their impact on carbon storage. *Nature Climate Change*, 4(9), 806–810. https://doi.org/10.1038/nclimate2318
- Seidl, R., Thom, D., Kautz, M., Martin-Benito, D., Peltoniemi, M., Vacchiano, G., et al. (2017). Forest disturbances under climate change. Nature Climate Change, 7(6), 395–402. https://doi.org/10.1038/nclimate3303
- Senf, C., & Seidl, R. (2021a). Mapping the forest disturbance regimes of Europe. *Nature Sustainability*, 4(1), 63–70. https://doi.org/10.1038/ s41893-020-00609-y

- Senf, C., & Seidl, R. (2021b). Storm and fire disturbances in Europe: Distribution and trends. *Global Change Biology*, 27(15), 3605–3619. https:// doi.org/10.1111/gcb.15679
- Sitch, S., Smith, B., Prentice, I. C., Arneth, A., Bondeau, A., Cramer, W., et al. (2003). Evaluation of ecosystem dynamics, plant geography and terrestrial carbon cycling in the LPJ dynamic global vegetation model. *Global Change Biology*, 9(2), 161–185. https://doi.org/10.1046/j.1365-2486.2003.00569.x
- Smith, B., Prentice, I. C., & Sykes, M. T. (2001). Global Ecology & Biogeography, 10, 621–637. http://dx.doi.org/10.1046/j.1466-822X.2001. t01-1-00256.x
- Smith, B., Wårlind, D., Arneth, A., Hickler, T., Leadley, P., Siltberg, J., & Zaehle, S. (2014). Implications of incorporating N cycling and N limitations on primary production in an individual-based dynamic vegetation model. *Biogeosciences*, 11(7), 2027–2054. https://doi.org/10.5194/ bg-11-2027-2014
- Smith, P., Gregory, P. J., van Vuuren, D., Obersteiner, M., Havlík, P., Rounsevell, M., et al. (2010). Competition for land. *Philosophical Transac*tions of the Royal Society B: Biological Sciences, 365(1554), 2941–2957. https://doi.org/10.1098/rstb.2010.0127
- Smyth, C. E., Smiley, B. P., Magnan, M., Birdsey, R., Dugan, A. J., Olguin, M., et al. (2018). Climate change mitigation in Canada's forest sector: A spatially explicit case study for two regions. *Carbon Balance and Management*, 13(1), 11. https://doi.org/10.1186/s13021-018-0099-z
- Smyth, C. E., Stinson, G., Neilson, E., Lemprière, T. C., Hafer, M., Rampley, G. J., & Kurz, W. A. (2014). Quantifying the biophysical climate change mitigation potential of Canada's forest sector. *Biogeosciences*, 11(13), 3515–3529. https://doi.org/10.5194/bg-11-3515-2014
- Spinoni, J., Vogt, J. V., Naumann, G., Barbosa, P., & Dosio, A. (2018). Will drought events become more frequent and severe in Europe? International Journal of Climatology, 38(4), 1718–1736. https://doi.org/10.1002/joc.5291
- Stajic, B., Zlatanov, T., Velichkov, I., Dubravac, T., & Trajkov, P. (2009). Past and recent coppice forest management in some regions of South Eastern Europe. Silva Balcanica, 10(10), 9–19.
- Suzuki, S. N., Tsunoda, T., Nishimura, N., Morimoto, J., & Suzuki, J. I. (2019). Dead wood offsets the reduced live wood carbon stock in forests over 50 years after a stand-replacing wind disturbance. *Forest Ecology and Management*, 432, 94–101. https://doi.org/10.1016/j. foreco.2018.08.054
- Taylor, K. E., Stouffer, R. J., & Meehl, G. A. (2012). An overview of CMIP5 and the experiment design. Bulletin of the American Meteorological Society, 93(4), 485–498. https://doi.org/10.1175/BAMS-D-11-00094.1
- Thomas, F. M., Rzepecki, A., & Werner, W. (2022). Non-native Douglas fir (Pseudotsuga menziesii) in Central Europe: Ecology, performance and nature conservation. *Forest Ecology and Management*, 506, 119956. https://doi.org/10.1016/j.foreco.2021.119956
- Thonicke, K., Venevsky, S., Sitch, S., & Cramer, W. (2001). The role of fire disturbance for global vegetation dynamics: Coupling fire into a dynamic global vegetation model. *Global Ecology and Biogeography*, 10(6), 661–677. https://doi.org/10.1046/j.1466-822X.2001.00175.x
- Uhde, B., Hahn, A., Griess, V. C., & Knoke, T. (2015). Hybrid MCDA methods to integrate multiple ecosystem services in forest management planning: A critical review. *Environmental Management*, 56(2), 373–388. https://doi.org/10.1007/s00267-015-0503-3
- Uhde, B., Heinrichs, S., Stiehl, C. R., Ammer, C., Müller-Using, B., & Knoke, T. (2017). Bringing ecosystem services into forest planning Can we optimize the composition of Chilean forests based on expert knowledge? *Forest Ecology and Management*, 404, 126–140. https://doi. org/10.1016/j.foreco.2017.08.021

United Nations. (2015a). Summary of the Paris Agreement. Retrieved from http://bigpicture.unfccc.int/%23content%2Dthe%2Dparis%2Dagreement. Retrieved from http://bigpicture.unfccc.int%20content%2Dthe%2Dparis%2Dagreement. Retrieved from http://bigpicture.unfccc.int%20content%2Dthe%2Dparis%2Dthe%2Dparis%2Dthe%2Dparis%2Dthe%2Dparis%2Dthe%2Dparis%2Dthe%2Dparis%2Dthe%2Dparis%2Dthe%2Dparis%2Dthe%2Dparis%2Dthe%2Dparis%2Dthe%2Dparis%2Dthe%2Dparis%2Dthe%2Dparis%2Dthe%2Dparis%2Dthe%2Dparis%2Dthe%2Dparis%2Dthe%2Dparis%2Dthe%2Dparis%2Dthe%2Dparis%2Dthe%2Dthe%2Dthe%2Dthe%2Dthe%2Dthe%2Dthe%2Dparis%2Dthe%2D

United Nations. (2015b). Transforming our world: The 2030 agenda for sustainable development. Retrieved from https://sdgs.un.org/2030agenda Vacchiano, G., Garbarino, M., Lingua, E., & Motta, R. (2017). Forest dynamics and disturbance regimes in the Italian Apennines. *Forest Ecology*

- and Management, 388, 57–66. https://doi.org/10.1016/j.foreco.2016.10.033 van Kooten, G. C., & Johnston, C. M. (2016). The economics of forest carbon offsets. Annual Review of Resource Economics, 8(1), 227–246. https://doi.org/10.1146/annurev-resource-100815-095548
- Verlinden, M. S., Broeckx, L. S., & Ceulemans, R. (2015). First vs. second rotation of a poplar short rotation coppice: Above-ground biomass productivity and shoot dynamics. *Biomass and Bioenergy*, 73, 174–185. https://doi.org/10.1016/j.biombioe.2014.12.012
- Virtanen, P., Gommers, R., Oliphant, T. E., Haberland, M., Reddy, T., Cournapeau, D., et al. (2020). SciPy 1.0: Fundamental algorithms for scientific computing in Python. *Nature Methods*, 17(3), 261–272. https://doi.org/10.1038/s41592-019-0686-2
- Vuidot, A., Paillet, Y., Archaux, F., & Gosselin, F. (2011). Influence of tree characteristics and forest management on tree microhabitats. *Biolog*ical Conservation, 144(1), 441–450. https://doi.org/10.1016/j.biocon.2010.09.030
- Winckler, J., Reick, C. H., & Pongratz, J. (2017). Robust identification of local biogeophysical effects of land-cover change in a global climate model. *Journal of Climate*, 30(3), 1159–1176. https://doi.org/10.1175/JCLI-D-16-0067.1
- Winckler, J., Reick, C. H., Bright, R. M., & Pongratz, J. (2019). Importance of surface roughness for the local biogeophysical effects of deforestation. Journal of Geophysical Research: Atmospheres, 124(15), 8605–8618. https://doi.org/10.1029/2018JD030127
- Wramneby, A., Smith, B., & Samuelsson, P. (2010). Hot spots of vegetation-climate feedbacks under future greenhouse forcing in Europe. Journal of Geophysical Research, 115(D21), D21119. https://doi.org/10.1029/2010JD014307

References From the Supporting Information

- Boisier, J. P., De Noblet-Ducoudré, N., & Ciais, P. (2013). Inferring past land use-induced changes in surface albedo from satellite observations: A useful tool to evaluate model simulations. *Biogeosciences*, 10(3), 1501–1516. https://doi.org/10.5194/bg-10-1501-2013
- Clarke, L., Edmonds, J., Jacoby, H., Pitcher, H., Reilly, J., & Richels, R. (2007). Scenarios of greenhouse gas emissions and atmospheric concentrations. Sub-report 2.1a of Synthesis and Assessment Product 2.1 by the US Climate Change Science Program and the Subcommittee on Global Change Research. Department of Energy, Office of Biological & Environmental Research.
- Dobor, L., Hlásny, T., Rammer, W., Zimová, S., Barka, I., & Seidl, R. (2020). Is salvage logging effectively dampening bark beetle outbreaks and preserving forest carbon stocks? *Journal of Applied Ecology*, 57(1), 67–76. https://doi.org/10.1111/1365-2664.13518
- Forrest, M., Tost, H., Lelieveld, J., & Hickler, T. (2020). Including vegetation dynamics in an atmospheric chemistry-enabled general circulation model: Linking LPJ-GUESS (v4.0) with the EMAC modelling system (v2.53). *Geoscientific Model Development*, 13(3), 1285–1309. https://doi.org/10.5194/gmd-13-1285-2020
- Fujino, J., Nair, R., Kainuma, M., Masui, T., & Matsuoka, Y. (2006). Multi-gas mitigation analysis on stabilization scenarios using aim global model. *The Energy Journal*, 27(01), 343–353. https://doi.org/10.5547/ISSN0195-6574-EJ-VolSI2006-NoSI3-17

Harja, D., Subekti, R., & Pambudi, S. (2019). Tree functional attributes and ecological database. Retrieved from http://db.worldagroforestry.org Hickler, T., Prentice, I. C., Smith, B., Sykes, M. T., & Zaehle, S. (2006). Implementing plant hydraulic architecture within the LPJ dynamic global vegetation model. *Global Ecology and Biogeography*, 15(6), 567–577. https://doi.org/10.1111/j.1466-8238.2006.00254.x



Hijioka, Y., Matsuoka, Y., & Nishimoto, H. (2008). Global GHG emission scenarios under GIC; concentration stabilization targets. *Journal of Global Environmental Engineering*, 13, 97–108.

Jenkins, J. C. (2004). Comprehensive database of diameter-based biomass regressions for North American tree species (Vol. 319). United States Department of Agriculture, Forest Service.

- Jones, H. G. (2013). Plants and microclimate (Vol. 22, No. 2). Cambridge University Press. https://doi.org/10.1017/CBO9780511845727
- Kärhä, K., Anttonen, T., Poikela, A., Palander, T., Laurén, A., Peltola, H., & Nuutinen, Y. (2018). Evaluation of salvage logging productivity and costs in windthrown Norway spruce-dominated forests. *Forests*, 9(5), 280. https://doi.org/10.3390/f9050280

Knoke, T., Gosling, E., Thom, D., Chreptun, C., Rammig, A., & Seidl, R. (2021). Economic losses from natural disturbances in Norway spruce forests – A quantification using Monte-Carlo simulations. *Ecological Economics*, 185, 107046. https://doi.org/10.1016/j.ecolecon.2021.107046 Kucharik, C. J., Norman, J. M., & Gower, S. T. (1998). Measurements of branch area and adjusting leaf area index indirect measurements. *Agri-*

cultural and Forest Meteorology, 91(1–2), 69–88. https://doi.org/10.1016/S0168-1923(98)00064-1 Lindenmayer, D., & Noss, R. (2006). Salvage logging, ecosystem processes, and biodiversity conservation. *Conservation Biology*, 20(4), 949–958. https://doi.org/10.1111/j.1523-1739.2006.00497.x

Lindenmayer, D. B., Burton, P. J., & Franklin, J. F. (2012). Salvage logging and its ecological consequences. Island Press.

- McGroddy, M. E., Daufresne, T., & Hedin, L. O. (2004). Scaling of C:N:P stoichiometry in forests worldwide: Implications of terrestrial redfield-type ratios. *Ecology*, 85(9), 2390–2401. https://doi.org/10.1890/03-0351
- Ministry of Agriculture and Forestry Finland. (2013). Forest Damages Prevention Act (1087/2013). Retrieved from http://www.finlex.fi/en/laki/kaannokset/2013/en20131087.pdf

Möllmann, T. B., & Möhring, B. (2017). A practical way to integrate risk in forest management decisions. *Annals of Forest Science*, 74(4), 75. https://doi.org/10.1007/s13595-017-0670-x

RCP Database. (2021). RCP database (version 2.0). Retrieved from https://tntcat.iiasa.ac.at/RcpDb

- Riahi, K., Grübler, A., & Nakicenovic, N. (2007). Scenarios of long-term socio-economic and environmental development under climate stabilization. *Technological Forecasting and Social Change*, 74(7), 887–935. https://doi.org/10.1016/j.techfore.2006.05.026
- Savill, P. S. (Ed.). (2019). The silviculture of trees used in British forestry. CABI. https://doi.org/10.1079/9781786393920.0000
- Smith, S. J., & Wigley, T. (2006). Multi-gas forcing stabilization with minicam. The Energy Journal, SI2006(01), 373–391. https://doi. org/10.5547/ISSN0195-6574-EJ-VolSI2006-NoSI3-19
- Stimm, B., Roloff, A., Lang, U. M., & Weisgerber, H. (Eds.). (2014). Enzyklopädie der Holzgewächse: Handbuch und Atlas der Dendrologie. Wiley. https://doi.org/10.1002/9783527678518
- Thorn, S., Chao, A., Georgiev, K. B., Müller, J., Bässler, C., Campbell, J. L., et al. (2020). Estimating retention benchmarks for salvage logging to protect biodiversity. *Nature Communications*, 11(1), 1–8. https://doi.org/10.1038/s41467-020-18612-4
- van Vuuren, D. P., den Elzen, M. G. J., Lucas, P. L., Eickhout, B., Strengers, B. J., van Ruijven, B., et al. (2007). Stabilizing greenhouse gas concentrations at low levels: An assessment of reduction strategies and costs. *Climatic Change*, 81(2), 119–159. https://doi.org/10.1007/ s10584-006-9172-9
- Wang, Z., & Zeng, X. (2010). Evaluation of snow albedo in land models for weather and climate studies. Journal of Applied Meteorology and Climatology, 49(3), 363–380. https://doi.org/10.1175/2009JAMC2134.1

West, P. W. (2014). Growing plantation forests (Vol. 9783319018). Springer International Publishing. https://doi.org/10.1007/978-3-319-01827-0
Wise, M., Calvin, K., Thomson, A., Clarke, L., Bond-Lamberty, B., Sands, R., et al. (2009). Implications of limiting CO₂ concentrations for land use and energy. Science, 324(5931), 1183–1186. https://doi.org/10.1126/science.1168475